

03.01.01 Ergon Energy's Building Block Components



Revision history

Version	Date	Summary of changes
1.0	31 October 2014	As submitted to the AER as part of the initial Regulatory Proposal
2.0	3 July 2015	As submitted to the AER as part of the revised Regulatory Proposal

Contents

Re	visior	n hist	ory	1
1	Intro	oduc	tion	4
	1.1	Ove	erview	4
	1.2	Pur	pose of this document	4
	1.3	NEF	R requirements	5
	1.4	Stru	Icture of this document	6
2	Reg	gulato	bry Asset Base	8
	2.1	Roll	forward of the RAB in the regulatory control period 2010-15	8
	2.1.	1	Opening RAB as at 1 July 2010	10
	2.1.	2	Capital expenditure during the current regulatory control period	10
	2.1.	3	Regulatory depreciation	13
	2.1. exp		Adjustment to recognise the difference between actual and forecast net capital ture in 2009-10	13
	2.1.	5	Adjustments to recognise changes in service classifications that occur on 1 July 2	
	2.2	Roll	forward of the RAB in the regulatory control period 2015-20	15
	2.2.	1	Capital expenditure	16
	2.2.	2	Regulatory depreciation	17
2	2.3	Roll	forward of the tax asset base in the regulatory control period 2010-15	18
	2.3.	1	Opening tax asset base as at 1 July 2010	19
	2.3.	2	Capital expenditure during the current regulatory control period	19
	2.3.	3	Tax depreciation	20
	2.3.	4	Adjustments to recognise changes in service classifications that occur on 1 July 2	
3	Ret	urn c	on capital	22
4	Ret	urn c	of Capital (Depreciation)	23
4	4.1	Stra	aight Line Depreciation	24
4	1.2	Ass	et lives	26
	4.2.	1	Standard Lives	26
	4.2.	2	Remaining Lives	26
	4.2.	3	Tax Standard Lives	31
	4.2.	4	Tax Remaining Lives	31
2	4.3	Inde	exation	31
5	Ope	eratin	ng expenditure	33

	5.1	Network Operating and Maintenance costs and Other O&M costs	33
	5.2	Debt raising costs	33
	5.3	Revenue Adjustments	34
6	Cor	porate income tax	35
	6.1	Departures	36
	6.1.		
		omer connections in 2015-20	
7	Rev	enue increments/decrements	
	7.1	Incentive schemes	
	7.1.	1 Efficiency Benefit Sharing Scheme	10
	7.1.	2 Service Target Performance Incentive Scheme (STPIS)	11
	7.1.3	3 Demand Management Incentive Scheme	12
	7.1.4	4 Capital Expenditure Sharing Scheme	13
	7.1.	5 Small Scale Incentive Scheme	13
	7.2	Adjustments from the application of the control mechanisms in the current regulatory	
		I period	
	7.2.		
	7.2.2	5	
	7.2.3		
	7.2.4	•	
	7.2.		
	7.2.	6 Cost Pass-Through – Feed-in tariffs	15
	7.3	Shared assets	
8	Ann	ual revenue requirement	18
	8.1	Comparison to forecast annual revenue requirement for regulatory control period 2010-1	
	8.2	Departures	50
9	X fa	ctors	51
	9.1	Relevant NER requirements	51
	9.2	X Factor Calculations – Standard Control Services	51
1	0 Li	st of documents referenced	55

1 Introduction

1.1 Overview

This document provides supporting information and explanations of the relevant components comprising the Standard Control Services Annual Revenue Requirement (ARR) building blocks for the regulatory control period 2015-20. It also considers how the actual and estimated expenditures, disposals, capital contributions and other allowable adjustments within the regulatory control period 2010-15 impact the opening asset base and ARR values for the regulatory control period 2015-20.

The approach the Australian Energy Regulator (AER) must take in determining the revenue requirements for Standard Control Services is detailed in Part C of Chapter 6 of the National Electricity Rules (NER).

To assist the AER determine the revenue requirements, we are required to develop a building block proposal. Building blocks, added together, allow the AER to determine the ARR for each regulatory year.¹

Our revised building block proposal contains the necessary information to allow the AER to make relevant decisions in accordance with the NER requirements. This document has been updated in response to the AER's Preliminary Determination and provides details of our revised building block components, specifies values derived for each component, and builds on the summary provided in our revised Regulatory Proposal. It also includes information on the X factors applied to the building block revenue.

We populated the AER's Post Tax Revenue Model (PTRM) with the information necessary for the AER to determine the ARR, including the revenue increments and decrements set out in clause 6.4.3 of the NER.²

1.2 Purpose of this document

This document sets out components of the revenue building block calculations that underpin our revised Regulatory Proposal. It provides detailed descriptions of, and justifications for, the assumptions, calculations and data that comprise the following inputs into the calculation of the ARR:

- roll forward of the Regulatory Asset Base (RAB) and tax asset base in the regulatory control period 2010-15, including:
 - o actual and estimated capital expenditure
 - o actual and estimated disposals
 - o actual and estimated capital contributions and gifted assets
 - o actual and estimated adjustments to the RAB
- roll forward of the RAB and tax asset base in the regulatory control period 2015-20, including:

¹ NER clause 6.4.3.

² Clause 6.4.2 of the NER requires the PTRM to set out how the ARR is to be determined. Further, clause 6.4.3 of the NER defines the building blocks that make up the ARR. We have interpreted these two clauses to mean the PTRM must include all building blocks set out in clause 6.4.3.

- o forecast capital expenditure
- o forecast disposals
- o forecast capital contributions and gifted assets
- return on asset calculations for the regulatory control period 2015-20
- regulatory depreciation and inflation additions for the regulatory control period 2015-20
- asset lives and tax asset lives (including standard and remaining lives) for the regulatory control period 2015-20
- forecast operating expenditure for the regulatory control period 2015-20
- the corporate income tax allowance for the regulatory control period 2015-20
- other revenue adjustments for the regulatory control period 2015-20, including:
 - o shared asset adjustments
 - o Efficiency Benefit Sharing Scheme (EBSS) revenue adjustments
 - o Service Target Performance Incentive Scheme (STPIS) revenue adjustments
 - o Demand Management Incentive Scheme (DMIS) revenue adjustments
 - o carry forward of Distribution Use of System (DUOS) over- and under-recoveries
- X factor calculations for the regulatory control period 2015-20.

1.3 NER requirements

Clause 6.4.3(a) of the NER sets out the building blocks that comprise the ARR:

- indexation of the regulatory asset base, with the regulatory asset base to be calculated in accordance with the NER, and a negative adjustment for inflation for that year
- a return on capital for that year, calculated by applying a rate of return for that regulatory year that is determined in accordance with the allowed rate of return to the value of the regulatory asset base
- the depreciation for that year, calculated on the value of the assets as included in the regulatory asset base, as at the beginning of that regulatory year
- our estimated cost of corporate income tax, estimated in accordance with the formula set out under the NER
- the revenue increments or decrements (if any) for that year arising from the application of any EBSS, Capital Expenditure Sharing Scheme (CESS), STPIS, Demand Management and Embedded Generation Connection Incentive Scheme or Small-scale Incentive Scheme (SSIS)
- the other revenue increments or decrements (if any) for that year arising from the application
 of a control mechanism in the previous regulatory control period. That is, revenue increments
 or decrements that are to be carried forward to the current regulatory control period as a result
 of the application of a control mechanism in the previous regulatory control period, and are
 apportioned to the relevant year under the distribution determination for the current regulatory
 control period
- the revenue decrements (if any) for that year arising from the use of assets that provide Standard Control Services to provide certain other services, that is as a result of those assets

being used to provide distribution services that are not classified as a direct control service or a negotiated distribution service; or services that are neither distribution services nor services that are provided by means of, or in connection with, dual function assets

• The forecast operating expenditure for that year, that is the forecast operating expenditure as accepted or substituted by the AER.

The diagram below sets out the high level relationships and interdependencies of the ARR.



1.4 Structure of this document

The remainder of this document is structured as follows:

- Section 2 sets out how the opening RAB value as of 1 July 2015 was established, having regard for:
 - o the opening RAB at the start of the regulatory control period 2010-15
 - actual and estimated capital expenditure, capital contributions and disposals during the regulatory control period 2010-15
 - regulatory depreciation during the regulatory control period 2010-15

 adjustments made during the regulatory control period 2010-15 to both remove and include assets in the RAB

Section 2 also describes the roll forward of the RAB during the regulatory control period 2015-20, having regard for:

- forecast capital expenditure, capital contributions and disposals during the regulatory control period 2015-20
- o regulatory depreciation during the regulatory control period 2015-20
- Section 3 summarises the method by which the return on capital was determined for the regulatory control period 2015-20
- Section 4 sets out how regulatory depreciation has been determined for the regulatory control period 2015-20, having regard for:
 - the calculation of remaining asset lives by asset class
 - o indexation of the RAB
- Section 5 summarises the method by which the operational expenditure was determined for the regulatory control period 2015-20
- Section 6 summarises the method by which the corporate income tax was estimated for the regulatory control period 2015-20, as well as the roll forward of the tax asset base to 1 July 2015
- Section 7 sets out how the various other adjustments, increments and decrements to the RAB and the ARR were determined for the regulatory control period 2015-20, having regard for:
 - adjustments from the application of the control mechanisms in the current regulatory control period (e.g. the carry forward of DUOS over- and under-recoveries)
 - incentive schemes
 - o shared assets
- Section 8 summarises the method by which the ARR was determined for the regulatory control period 2015-20
- Section 9 outlines how the X factors for the Standard Control Services ARR values were developed.

2 Regulatory Asset Base

In accordance with Schedule 1 clause 27.1 of the Reset Regulatory Information Notice (RIN), this chapter summarises the establishment of the 1 July 2015 opening RAB value and specifically covers:

- The calculation of the RAB for each year of the regulatory control period 2010-15, including:
 - the opening RAB at the start of the regulatory control period 2010-15
 - the actual and estimated capital expenditure, capital contributions and disposals during the regulatory control period 2010-15
 - o the regulatory depreciation during the regulatory control period 2010-15
 - the other adjustments made to the RAB during the regulatory control period 2010-15 to recognise departures to the underlying methods in the AER's Roll Forward Model (RFM) and Guidelines.

2.1 Roll forward of the RAB in the regulatory control period 2010-15

In its Preliminary Determination, the AER proposed an opening RAB at 1 July 2015 of \$10,102.2 million. Ergon Energy does not agree with the AER's proposed opening RAB. However, we do accept the following aspects of the AER's Preliminary Determination and have adjusted our opening RAB as at 1 July 2015 accordingly:

- We accept the AER's decision to remove the Hayman Island undersea cable assets in the opening RAB as of 1 July 2015.
- We accept the AER's decision to include the approved equity raising cost for the regulatory control period 2010-15 in 2011-12 rather than 2010-11 as per our October Regulatory Proposal.
- We accept the AER's decision to remove the movement in capitalised provisions from the RFM.
- We accept the AER's decision to replace the actual and estimated disposals set out in our Regulatory Proposal with updated disposals value provided to the AER as part of an information request subsequent to the Regulatory Proposal submission.

The remainder of this section provides justifications for the establishment our opening RAB of \$10,055.8 million in the revised Regulatory Proposal.

Our RAB represents the remaining value of all previous capital investments to be recovered from customers, taking into account:

- investment already recovered from customers through the depreciation allowance
- investment in new assets
- any proceeds from asset disposals
- increases or decreases in the value of previous investments due to the asset providing a different service or classification of service.

We have rolled forward our opening RAB to 1 July 2015 in accordance with:

- Schedule 2.1 of the NER, which establishes how our opening RAB must be calculated
- the relevant transitional provisions set out in clause 6.5.1 of the NER
- the AER's RFM and Guidelines, which set out how our prior and future investments are incorporated into our revenue requirement.

This is because the detailed calculations underpinning the establishment of the opening RAB as of 1 July 2015 are embodied within the AER's RFM (consistent with the requirements of Schedule 1 clause 27.1 of the Reset RIN). The estimated value of our RAB at 1 July 2015 of \$10,055.8 million was calculated in accordance with clause 6.5.1 and Schedule 6.2 of the NER and comprises:

- the roll forward of actual capital expenditure for 2010-11 to 2013-14
- revised estimated capital expenditure for 2014-15
- adjustments to reflect the changes in service classifications as of 1 July 2015.

Whilst we have not altered the calculations underpinning the AER's RFM, we have made some departures from the underlying inputs and methods in the AER's RFM. These relate to:

- the treatment of capital contributions (as set out in section 2.1.2.2)
- the assets which are being included or removed from the RAB during the regulatory control period 2010-15 due to service reclassifications (as set out in section 2.1.5).

In accordance with Schedule 1 clause 27.2 of the Reset RIN, these departures and the reasons underpinning them are set out in the appropriate sections within this chapter. We note that all departures have been correctly applied in the context of the AER's 2010-15 Distribution Determination and the NER in effect at the time.

The table below sets out the annual roll-forward of the Standard Control Services RAB for each regulatory year of the regulatory control period 2010-15.

	2010-11	2011-12	2012-13	2013-14	2014-15
Opening RAB	7,148.95	7,843.82	8,375.96	9,034.88	9,649.23
<i>plus</i> Capital Expenditure (net of disposals and capital contributions)	809.48	758.18	827.97	744.00	799.60
less Regulatory depreciation	(114.61)	(226.04)	(169.05)	(129.65)	(120.59)
<i>less</i> Difference Between Actual and Forecast Net Capital Expenditure in 2009- 10, and the Return on Difference for the Net Capital Expenditure in 2009- 10	-	-	-	-	(210.80)
Closing RAB	7,843.82	8,375.96	9,034.88	9,649.23	10,117.43

Table 1: Ergon Energy's Regulatory Asset Base for 2010-15 (\$M nominal)

	2010-11	2011-12	2012-13	2013-14	2014-15
<i>less</i> adjustments to recognise changes in service classifications that occur on 1 July 2015	-	-	-	-	(61.60)
Opening RAB 1 July 2015					10,055.83

The closing RAB values reflect the values in the RFM for the regulatory control period 2010-15. The opening RAB value as of 1 July 2015 reflects the value in the Standard Control Service PTRM for the regulatory control period 2015-20.

2.1.1 Opening RAB as at 1 July 2010

The value of the RAB as at 1 July 2010 was calculated to be \$7,148.95 million. The value was determined as follows:

- an opening asset base of \$6,452.64 million as per the 2010-15 Distribution Determination
- plus the nominal forecast net capital expenditure for 2009-10 of \$819.46 million as per the 2010-15 Distribution Determination
- less the nominal forecast regulatory depreciation for 2009-10 of \$123.15 million as per the 2010-15 Distribution Determination.

2.1.2 Capital expenditure during the current regulatory control period

For the purposes of rolling forward the RAB from 1 July 2010 to the end of the regulatory control period 2010-15, the net capital expenditure needs to be determined. Net capital expenditure is the actual (or estimated) capital expenditure for a given regulatory year, less the capital contributions, disposals and other allowable adjustments (as per section 2.1.4 of this document) that occurred during that regulatory year.

The table below sets out the capital expenditure, capital contributions, disposals and the resulting net capital contributions for each year of the regulatory control period 2010-15. Actual values are provided for 2010-11 through to 2013-14 inclusive, with revised estimated values provided for 2014-15. Note that these values are adjusted by the RFM to include a half-year timing adjustment equal to the square root of (1 + vanilla Weighted Average Cost of Capital (WACC)) in each regulatory year of the regulatory control period 2010-15. We have not modified this adjustment within the RFM and we present these values in the table below.

	2010-11	2011-12	2012-13	2013-14	2014-15
Capital expenditure	821.80	842.70	836.00	758.75	806.63
less Capital contributions	-	-	-	-	-

Table 2: Ergon Energy's net capital expenditure 2010-15 (\$M nominal)

	2010-11	2011-12	2012-13	2013-14	2014-15
less Disposals	(12.32)	(84.52)	(8.03)	(14.75)	(10.03)
Net capital expenditure	809.48	758.18	827.97	744.00	799.60

2.1.2.1 Capital expenditure

The capital expenditure values for 2010-11 through to 2013-14 inclusive, prior to the half-year timing adjustment being applied, represent the actual annual total capital expenditure values reported in the Annual Reporting RINs. The revised capital expenditure value for 2014-15 has been developed based on the best estimate available at the time of preparing the revised Regulatory Proposal. A more detailed description of how the capital expenditure values for these years were developed is provided in Appendix B of the revised Regulatory Proposal and in the various capital expenditure summary documents that form part of the revised Regulatory Proposal submission.

2.1.2.2 Capital contributions

Clause 6.21 of the NER details the circumstances in which we can "*minimise financial risks* associated with investment in network assets and provides for adoption of cost reflective payment options in conjunction with the use of average distribution prices". In particular:

- clause 6.21.2(2) provides that we "may receive a capital contribution, prepayment and/or financial guarantee up to the provider's future revenue related to the provision of direct control services for any new assets installed as part of a new connection or modification to an existing connection, including any augmentation to the distribution network", and
- clause 6.21.2(3) provides that "where assets have been the subject of a contribution or prepayment, the Distribution Network Service Provider must amend the provider's revenue related to the provision of direct control services".

In addition, a Queensland-specific transitional rule was introduced for the regulatory control period 2010-15 under clause 11.16.10 of the NER. This clause provided that:

- By 1 July 2009, we must publish on our website a capital contributions policy based on the requirements relating to capital contributions in our Pricing Principles Statement (PPS) immediately in force prior to 1 July 2009.
- The AER may, before 1 January 2010, direct us in writing to revise and republish our capital contributions policy if it is inconsistent with our PPS.
- We may apply to the AER to amend our capital contributions policy after 1 January 2010.

The requirements of Chapters 6 and 11 of the NER meant that we were able to retain our existing capital contributions policy in the regulatory control period 2010-15 provided that it remained consistent with our approved PPS. The existing methodology was approved by the Queensland Competition Authority in 2005 and involved contributed assets being recognised in the RAB at their full value, but the return on and return of capital associated with these contributed assets was netted off the ARR each year.

The forecast capital contributions for 2014-15 are based on the best estimate available at the time of preparing the revised Regulatory Proposal and the assumptions that capital contributions remain the same for the rest of the current regulatory control period. These estimated capital contributions, together with the actual capital contributions for 2010-11 to 2013-14 inclusive are included in the RAB roll forward for the current regulatory control period and have not been removed from the Standard Control Services RAB.

This represents a departure from the underlying methods of the AER's RFM (in accordance with Schedule 1 clause 27.1 of the Reset RIN). This departure has been adopted because the transitional arrangements in clause 11.16.10 of the NER allow the RAB used to determine the allowable revenue for the regulatory control period 2010-15 to include a value for the forecast capital contributions (being both cash and gifted assets). Therefore, the calculated revenue included an allowance for return of, and on, the contributed assets.

To offset the revenue resulting from the inclusion of capital contributions, a revenue adjustment equal to the value of the forecast capital contributions was applied in the year in which the capital contribution was forecast to occur. By definition, the net present value (NPV) of the revenue stream to be earned from the capital contributions over the life of those assets is equal to the initial value of the capital contribution. A conceptual illustration of this mechanism is provided in the diagram below.



As illustrated in the diagram, the capital contributions are not removed from the RAB as doing so would result in the NPV of the revenue stream from those assets being lower than the original value of the contributions (i.e. the original revenue adjustment would have been too high). Therefore, the value of the actual capital contributions for the current regulatory control period have been included in the roll forward of the RAB to 1 July 2015, so that the forward revenue calculations will continue to include an amount for the return on, and of, the past capital contributions.

2.1.2.3 Disposals

Clause S6.2.1(e)(6) of the NER requires the RAB to be reduced by the 'disposal value' of assets. We note that 'disposal value' is not defined in the NER. However, section 5.3.2 of the AER's RFM Final Decision for Electricity Distribution Network Service Providers (June 2008) states that:

"For the purposes of the RFM, the AER accordingly considers that using the sale or depreciated value as the disposal value of an asset may be acceptable. The AER will assess the appropriateness of either of these approaches as proposed by a DNSP on a case-by-case basis. In either case the AER also notes that the approach adopted in the RFM must be consistent with that applied on a forecast basis in the PTRM."

We have adopted the sale value of assets to be the disposal value in both the RFM (for the regulatory control period 2010-15) and for the PTRM (for the regulatory control period 2015-20). This approach is consistent with the principles of Financial Capital Maintenance and is consistent with what we proposed and was approved by the AER in the 2010-15 Distribution Determination.

The disposal amounts reflect:

- the transfer of the Tarong to Columboola 132kV line and the Chinchilla and Columboola 132kV switchyards to Powerlink in 2011-12
- the disposal of motor vehicles
- the disposal of other non-system assets, including three specific corporate disposals identified by the AER in its Preliminary Determination.

2.1.3 Regulatory depreciation

A detailed discussion on the approach taken to determine regulatory depreciation for the regulatory control period 2010-15 and the regulatory control period 2015-20 is set out in section 4 of this document.

2.1.4 Adjustment to recognise the difference between actual and forecast net capital expenditure in 2009-10

The AER's RFM includes a mechanism for recognising the difference between forecast and actual capital expenditure in 2009-10 in the roll forward of the RAB in the regulatory control period 2010-15. This is because at the time of the 2010-15 Distribution Determination being prepared, the actual capital expenditure for 2009-10 was not available. The RFM determines the difference between the actual and forecast capital expenditure in 2009-10 and deducts the difference from the closing 2014-15 RAB. It also deducts the return on the difference between actual and forecast capital expenditure for 2014-15 RAB.

We have applied this mechanism in the RFM without modification as follows to obtain a total adjustment in 2014-15 of \$210.80 million, which comprises the sum of:

- The difference between the nominal actual net capital expenditure (\$690.33 million) and nominal forecast net capital expenditure in 2009-10 (\$819.46 million) was calculated to be \$131.95 million. The nominal actual net capital expenditure of \$690.33 million comprises:
 - \$673.10 million of nominal actual capital expenditure incurred in 2009-10
 - o less \$11.56 million of nominal actual disposals incurred in 2009-10

plus \$28.78 million of timing adjustments as calculated by the AER's RFM. The AER's RFM does this by multiplying the actual nominal net capital expenditure by the square root of (1 + the nominal vanilla WACC) for the previous regulatory control period (8.89%)

The revenue benefit arising from this capital expenditure underspend of \$131.95 million has been returned to customers. This is because in accordance with the RFM, we only include the actual net capital expenditure in the 2010-11 opening RAB and hence only earn the depreciation and return on asset allowance on the actual net capital expenditure from 2010-11 onwards.

 The return on the difference between actual and forecast capital expenditure is calculated to be \$78.85 million, which is the sum of the compounded return on the difference in each year of the regulatory control period 2010-15. This calculation is embodied in the Adjustment for Previous Period tab of the AER's RFM.

2.1.5 Adjustments to recognise changes in service classifications that occur on 1 July 2015

We own an undersea cable to Hayman Island which at the start of the regulatory control period 2010-15, was not included in the RAB for Standard Control Services. In accordance with the AER's Preliminary Determination, we will <u>not</u> be transferring this asset to the RAB for Standard Control Services at the commencement of the regulatory control period 2015-20.

In addition, there are Type 5-6 metering assets which at the start of the regulatory control period 2010-15 were included in the RAB for Standard Control Services but will be removed from the RAB for Standard Control Services at the commencement of the regulatory control period 2015-20. This is due to the reclassification of Type 5-6 metering services as an Alternative Control Service from 1 July 2015.

Table 2. Ennon Engrande not	adjusting and a to us a similar	akammaa in aamulaa	alagaitiantiana (CM naminal)
Lable 3: Froon Energy S net	adjustments to recoonise	changes in service	classifications (\$M nominal)
		onlangee in eer mee	

Asset	Standard Control Service RAB Asset Class affected	Value as at 1 July 2015
Type 5-6 metering	Metering	(61.60)
Net RAB adjustment		(61.60)

Consistent with Schedule 1 clause 27.3 of the Reset RIN, the following sections provide details on:

- the applicable NER requirements relating to RAB adjustments due to service reclassifications
- the assets which are being included or removed from the RAB as at 1 July 2015.

2.1.5.1 Relevant NER requirements

Clause S6.2.1(e)(7) of the NER requires that:

The previous value of the regulatory asset base must be reduced by the value of an asset where the asset was previously used to provide Standard Control Services (or their equivalent under the previous regulatory system) but, as a result of a change to the classification of a particular service under Part B, is not to be used for that purpose for the relevant regulatory control period.

And clause S6.2.1(e)(8) of the NER requires that:

The previous value of the regulatory asset base may be increased by the value of an asset to which this subparagraph applies to the extent that:

- *(i)* The AER considers the asset to be reasonably required to achieve one or more of the capital expenditure objectives, and
- (ii) The value of the asset has not been otherwise recovered.

This subparagraph applies to an asset that:

- (i) Was not used to provide Standard Control Services (or their equivalent under the previous regulatory system) in the previous regulatory control period but, as a result of a change to the classification of a particular service under Part B, is to be used for that purpose for the relevant regulatory control period; or
- (ii) Was never previously used to provide Standard Control Services (or their equivalent under the previous regulatory system) but is to be used for that purpose for the relevant regulatory control period.

The following sections set out the adjustments to the RAB being proposed and how these are in accordance with clause S6.2.1(e)(7) and clause S6.2.1(e)(8) of the NER.

2.1.5.2 Changes in RAB values due to service reclassifications - deductions

In accordance with the AER's Framework and Approach for Ergon Energy, Type 5-6 metering services will be classified as an Alternative Control Service as of 1 July 2015, representing a change from its previous classification as a Standard Control Service in the regulatory control period 2010-15. To recognise this reclassification, we have adjusted the Standard Control Services opening RAB in the PTRM for the regulatory control period 2015-20 by deducting the value of Type 5-6 meters (\$61.60 million) from the Metering asset class. This is consistent with the requirements of clause S6.2.1(e)(7) of the NER.

Further details on the treatment of Type 5-6 meters is provided in our summary document 05.03.01 - (Revised) Default Metering Services Summary which forms part of the revised Regulatory Proposal submission. This document also sets out the reasons for establishing a new separate asset class for Type 5-6 meters from 2015-16 onwards as required by Schedule 1 clause 28.4 of the Reset RIN. Consistent with the requirements of Schedule 1 clause 28.5 of the Reset RIN, we note that no other asset classes have been added, modified or removed in the asset base for the regulatory control period 2015-20.

2.2 Roll forward of the RAB in the regulatory control period 2015-20

This section summarises the roll forward of the opening RAB as of 1 July 2015 for the regulatory control period 2015-20. Specifically it considers the contribution of the following components to RAB roll forward:

• forecast capital expenditure for each regulatory year from 2015-16 to 2019-20

- forecast capital contributions for each regulatory year from 2015-16 to 2019-20, having regard for the cessation of the relevant transitional arrangements in place for the regulatory control period 2010-15
- forecast disposals for each regulatory year from 2015-16 to 2019-20.

We applied the AER's PTRM to roll forward the RAB for Standard Control Services from 1 July 2015 to 30 June 2020. A summary of the roll forward values is provided in Table 4.

	2015-16	2016-17	2017-18	2018-19	2019-20
Opening RAB	10,055.83	10,674.58	11,225.76	11,750.05	12,252.24
<i>plus</i> Capital Expenditure (net of disposals and capital contributions)	781.03	730.34	692.33	673.32	686.22
less Regulatory depreciation	(162.28)	(179.16)	(168.04)	(171.13)	(148.53)
Closing RAB	10,674.58	11,225.76	11,750.05	12,252.24	12,789.93

Table 4: Ergon Energy's RAB for 2015-20 (\$M nominal)

2.2.1 Capital expenditure

This section sets out how the manner in which forecast capital expenditure for the regulatory control period 2015-20 is recognised in the RAB. It has specific regard for the annual forecast capital expenditure, and the forecast capital contributions and forecast disposals which are netted from the annual capital expenditure values in each regulatory year. In addition, the forecast annual capital expenditures have been adjusted to reflect that some of the Standard Control Service non-system assets are also used in the provision of services other than Standard Control Services.

The table below sets out the forecast capital expenditure, capital contributions, disposals and the resulting net capital contributions for each year of the regulatory control period 2015-20. These values are included as inputs to the PTRM.

Consistent with our regulatory proposal submission, Ergon Energy is proposing equity raising costs in the revised Regulatory Proposal. This is explicitly set out in Appendix C of our revised Regulatory Proposal. Equity raising costs have been included in the forecast capital expenditure in 2015-16 and have been calculated using the methodology embodied within the AER's PTRM.

Table 5: Ergon Energy's net capital expenditure 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
Capital expenditure	922.32	882.98	857.12	858.61	877.11
less Capital contributions	(133.94)	(145.11)	(157.07)	(167.46)	(177.66)
less Disposals	(7.35)	(7.53)	(7.73)	(17.83)	(13.23)

	2015-16	2016-17	2017-18	2018-19	2019-20
Net capital expenditure	781.03	730.34	692.33	673.32	686.22

2.2.1.1 Capital expenditure

Our capital expenditure forecasts for the regulatory control period 2015-20, and the associated supporting justifications and explanations are set out in the Appendix B of the revised Regulatory Proposal.

2.2.1.2 Capital contributions

As noted in section 2.1.2.2, clause 6.21 of the NER details the circumstances in which we can *"minimise financial risks associated with investment in network assets and provides for adoption of cost reflective payment options in conjunction with the use of average distribution prices"*.

We note that the Queensland-specific transitional rule relating to our capital contributions policy will no longer apply from 1 July 2015 onwards. To reflect this, from 1 July 2015 we have deducted the annual forecast capital contributions (by asset class) from the annual forecast capital expenditure for each year of the regulatory control period. This has been achieved by entering the forecast capital contributions (by asset class) as input into the Input tab of the AER's PTRM in accordance with the PTRM Guidelines. The revenue deductions set out in section 2.1.2.2 are no longer applied for forecast capital contributions in the regulatory control period 2015-20.

Appendix B of the revised Regulatory Proposal and the Customer Connection Initiated Capital Works provide details on our forecast of capital expenditure which relates to customer connections, some of these works will involve upfront capital contributions or contributed assets. Our connection policy includes details of the circumstances in which some customers will pay a contribution towards their connection (or be responsible for the gifting of connection assets).

2.2.1.3 Disposals

As noted in section 2.1.2.3, we have adopted the sale value of assets to be the disposal value in both the RFM (for the regulatory control period 2010-15) and for the PTRM (for the regulatory control period 2015-20). This is consistent with the approach we proposed in our 2010-15 regulatory submission, which was approved by the AER in the 2010-15 Distribution Determination.

The disposals represent the forecast annual proceeds from sale of motor vehicles. Consistent with our approach to forecasting disposals in the 2010-15 regulatory submission, the forecast motor vehicle disposals for the regulatory control period 2015-20 are based on our historical proceeds from the sale of motor vehicles in real terms.

2.2.2 Regulatory depreciation

A detailed discussion on the approach taken to determine regulatory depreciation for the regulatory control period 2015-20 is set out in section 4 of this document.

2.3 Roll forward of the tax asset base in the regulatory control period 2010-15

This section summarises the roll forward of the opening tax asset base from 1 July 2010 to 1 July 2015 for the regulatory control period 2015-20. Note that the AER's PTRM does not roll forward the tax asset base for each year of the regulatory control period 2015-20 and as such this document does not cover the roll forward of the tax asset base beyond 1 July 2015.

Our tax asset base represents the remaining tax value of all previous capital investments to be recovered from customers, taking into account:

- the tax depreciation allowance
- investment in new assets
- any proceeds from asset disposals
- increases or decreases in the value of previous investments due to the asset providing a different service or classification of service.

We have rolled forward our opening tax asset base to 1 July 2015 in accordance with the AER's RFM and Guidelines, which set out how our prior and future investments are incorporated into our tax depreciation and tax payable calculations.

This is because the detailed calculations underpinning the establishment of the opening tax asset base as of 1 July 2015 are embodied within the AER's RFM (consistent with the requirements of Schedule 1 clause 29.6 of the Reset RIN). The estimated value of our tax asset base at 1 July 2015 of \$6,256.75 million comprises the roll forward of actual capital expenditure for 2010-11 to 2013-14, revised estimated capital expenditure for 2014-15 and adjustments to reflect the changes in service classifications as of 1 July 2015.

Whilst we have not altered the calculations underpinning the AER's RFM, we have made some departures from the underlying inputs and methods in the AER's RFM. These relate to the assets which are being included or removed from the tax asset base during the regulatory control period 2010-15 due to service reclassifications (as set out in section 2.1.5).

In accordance with Schedule 1 clause 29.7 of the Reset RIN, these departures and the reasons underpinning them are set out in the appropriate sections within this chapter. We note that all departures have been correctly applied in the context of the AER's 2010-15 Determination and the NER in effect at the time.

The table below sets out the annual roll-forward of the Standard Control Services tax asset base for each regulatory year of the regulatory control period 2010-15.

	2010-11	2011-12	2012-13	2013-14	2014-15
Opening Tax Asset Base	3,914.91	4,447.37	4,916.65	5,443.91	5,860.75
<i>plus</i> Capital Expenditure (net of disposals)	769.75	727.16	790.52	708.86	762.50

Table 6: Ergon Energy's Tax Asset Base for 2010-15 (\$M nominal)

	2010-11	2011-12	2012-13	2013-14	2014-15
<i>less</i> Tax Asset Base depreciation	(237.30)	(257.89)	(263.25)	(291.88)	(302.16)
Closing Tax Asset Base as at 30 June 2015	4,447.37	4,916.65	5,443.91	5,860.90	6,321.24
<i>less</i> adjustments to recognise changes in service classifications that occur on 1 July 2015	-	-	-	-	(64.49)
Opening Tax Asset Base as at 1 July 2015					6,256.75

The closing tax asset value reflects the values in the RFM for the regulatory control period 2010-15. The opening tax asset value as of 1 July 2015 reflects the value in the PTRM for the regulatory control period 2015-20.

2.3.1 Opening tax asset base as at 1 July 2010

The value of the tax asset base as at 1 July 2010 was calculated to be \$3,914.91 million. The value was determined as follows:

- an opening asset base for 2009-10 of \$3,458.29 million as per the 2010-15 Distribution Determination
- plus the nominal actual net capital expenditure for 2009-10 of \$661.55 million
- less the nominal actual tax depreciation for 2009-10 of \$204.92 million.

This value of the opening tax asset base on 1 July 2010 is different to the 2010-15 Distribution Determination opening tax asset base value of \$4,029.67 million because:

- Our actual net capital expenditure (i.e. net of disposals) in 2009-10 was less than forecast net capital expenditure for 2009-10 of \$786.21 million.
- Our actual tax depreciation in 2009-10 was lower than forecast depreciation for 2009-10 of \$214.83 million.

2.3.2 Capital expenditure during the current regulatory control period

For the purposes of rolling forward the tax asset base from 1 July 2010 to the end of the regulatory control period 2010-15, the net capital expenditure needs to be determined. Net capital expenditure for tax purposes is the actual (or estimated) capital expenditure for a given regulatory year, less disposals and other allowable adjustments (as per section 2.1.4 of this document) that occurred during that regulatory year.

The table below sets out the capital expenditure, disposals and the resulting net capital contributions for each year of the regulatory control period 2010-15. Actual values are provided for 2010-11 through to 2013-14 inclusive, with revised estimated values provided for 2014-15. Note that these values are <u>not</u> adjusted to include a half-year timing adjustment equal to the square root

of (1 + vanilla WACC) in each regulatory year of the regulatory control period 2010-15 (this is consistent with the AER's RFM). These values are presented in the table below.

	2010-11	2011-12	2012-13	2013-14	2014-15
Capital expenditure	781.47	808.22	798.19	722.92	772.07
less Disposals	(11.72)	(81.06)	(7.67)	(14.06)	(9.57)
Net capital expenditure	769.75	727.16	790.52	708.86	762.50

Table 7: Ergon Energy's net capital expenditure 2010-15 (\$M nominal)

2.3.2.1 Capital expenditure

The capital expenditure values for 2010-11 through to 2013-14 inclusive represent the actual annual total capital expenditure values reported in the Annual Reporting RINs. Refer to section 2.1.2.1 for further details.

2.3.2.2 Disposals

Refer to section 2.1.2.3 for further details.

2.3.3 Tax depreciation

A detailed discussion on the approach taken to determine tax depreciation for the regulatory control period 2010-15 and the regulatory control period 2015-20 is set out in section 4 of this document.

2.3.4 Adjustments to recognise changes in service classifications that occur on 1 July 2015

As noted in section 2.1.5, we own an undersea cable to Hayman Island which, at the start of the regulatory control period 2010-15, was not included in the tax asset base for Standard Control Services. In accordance with the AER's Preliminary Determination, we will not be transferring this asset to the tax asset base for Standard Control Services at the commencement of the regulatory control period 2015-20.

In addition, at the start of the regulatory control period 2010-15, Type 5-6 metering assets were included in the tax asset base for Standard Control Services. By virtue of the reclassification of Type 5-6 metering services as an Alternative Control Service from 1 July 2015, these assets will be removed from the tax asset base for Standard Control Services at the commencement of the regulatory control period 2015-20.

Table 8: Ergon Energy's net adjustments to recognise changes in service classifications (\$M nominal)

Asset	Standard Control Service Asset Class affected	Value as at 1 July 2015
Type 5-6 metering	Metering	64.49
Net tax asset base adjustment		-64.49

The following sections provide details on the assets which are being included or removed from the tax asset base as at 1 July 2015.

2.3.4.1 Changes in tax asset base values due to service reclassifications - deductions

In accordance with the AER's Framework and Approach and Preliminary Determination for Ergon Energy, Type 5-6 Metering will be classified as an Alternative Control Service as of 1 July 2015, representing a change from its previous classification as a Standard Control Service in the regulatory control period 2010-15. For this reason, we have adjusted the Standard Control Services opening tax asset base in the PTRM for the regulatory control period 2015-20 to recognise this reclassification by deducting the tax value of Type 5-6 meters (\$64.49 million) from the Metering asset class.

Further details on the treatment of Type 5-6 meters is provided in our supporting document 05.03.01 - (Revised) Default Metering Services Summary which forms part of the revised Regulatory Proposal submission.

3 Return on capital

An allowance for the return on capital is one of the building blocks which make up our ARR under clause 6.4.3(b)(2) of the NER. The return on capital is calculated as the product of the allowed rate of return and the opening value of the RAB used to provide Standard Control Services for that regulatory year.³

For the purposes of the building block calculation, the following WACC parameter inputs have been included in the PTRM:

WACC Parameter	Value
Corporate Tax Rate	30%
Inflation Rate	2.55%
Cost of Equity	10.00%
Utilisation of Imputation (Franking) Credits	25%
Proportion of Debt Funding	60%
Trailing Average Cost of Debt: 2015-16	5.68%
Debt Raising Cost Benchmark	0.197%

 Table 9: Ergon Energy's proposed WACC parameters

We have proposed departures to the Rate of Return Guidelines – refer to Appendix C of the revised Regulatory Proposal for details of, and justifications for the proposed departures.

³ NER clause 6.5.2(a)

4 Return of Capital (Depreciation)

Consistent with the requirements of Schedule 1 clause 28.1 of the Reset RIN, this chapter sets out our calculation of the Standard Control Services depreciation amounts for the regulatory control period 2010-15 and 2015-20.

The AER's June 2008 Electricity Distribution Network Service Providers RFM Handbook defines nominal regulatory depreciation as being nominal straight-line depreciation less the inflation applied to the opening RAB.

The calculation of nominal regulatory depreciation is embedded within the AER's RFM and PTRM, and in applying the RFM and PTRM we have calculated the nominal regulatory depreciation in a manner consistent with the definition set out in the RFM Handbook. Table 10 below sets out the regulatory depreciation, the straight line depreciation and the indexation components for each year of the regulatory control period 2010-15.

Table 10: Ergon Energy's regulatory depreciation for 2010-15 (\$M nominal)

	2010-11	2011-12	2012-13	2013-14	2014-15
Straight line depreciation	(352.67)	(349.98)	(378.45)	(394.37)	(385.95)
plus Indexation	238.06	123.93	209.40	264.72	265.35
Regulatory depreciation	(114.61)	(226.04)	(169.05)	(129.65)	(120.59)

Table 11 below sets out the tax depreciation for each year of the regulatory control period 2010-15.

Table 11: Ergon Energy's tax depreciation for 2010-15 (\$M nominal)

	2010-11	2011-12	2012-13	2013-14	2014-15
Tax depreciation	(237.30)	(257.89)	(263.25)	(291.88)	(302.16)

Table 12 below sets out the regulatory depreciation, and the straight line depreciation and indexation components, for each year of the regulatory control period 2015-20.

Table 12: Ergon Energy's regulatory depreciation for 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
Straight line depreciation	(418.70)	(424.81)	(404.67)	(395.34)	(360.85)
plus Indexation	256.42	248.87	240.68	232.50	224.07
Regulatory depreciation	(162.28)	(179.16)	(168.04)	(171.13)	(148.53)

Table 13 below sets out the tax depreciation for each year of the regulatory control period 2015-20.

	2015-16	2016-17	2017-18	2018-19	2019-20
Tax depreciation	(364.16)	(337.86)	(336.55)	(348.56)	(372.61)

Table 13: Ergon Energy's tax depreciation for 2015-20 (\$M nominal)

The sections below provide more detail on how the depreciation and inflation components of the regulatory depreciation building block were calculated for the regulatory control period 2010-15 and the regulatory control period 2015-20. Note that tax depreciation is calculated on a straight line basis using tax standard lives and tax remaining lives for each asset class. The straight line approach for tax is otherwise the same as that for the regulatory asset values and as such is not covered separately in this document.

4.1 Straight Line Depreciation

Clause 6.5.5 of the NER requires that:

- (a) The depreciation for each regulatory year:
 - (1) Must be calculated on the value of the assets as included in the RAB, as at the beginning of that regulatory year, for our distribution system, and
 - (2) Must be calculated:
 - (iii) Providing such depreciation schedules conform with the requirements set out in paragraph (b), using the depreciation schedules for each asset or category of assets that are nominated in our building block proposal, or
 - (iv) To the extent the depreciation schedules nominated in our building block proposal do not so conform, using the depreciation schedules determined for that purpose by the AER.

(b) The depreciation schedules referred to in paragraph (a) must conform to the following requirements:

- (1) The schedules must depreciate using a profile that reflects the nature of the assets or category of assets over the economic life of that asset or category of assets
- (2) The sum of the real value of the depreciation that is attributable to any asset or category of assets over the economic life of that asset or category of assets (such real value being calculated as at the time the value of that asset or category of assets was first included in the regulatory asset base for the relevant distribution system) must be equivalent to the value at which that asset or category of assets was first included in the regulatory asset base for the relevant distribution system
- (3) The economic life of the relevant assets and the depreciation methods and rates underpinning the calculation of depreciation for a given regulatory control period must be consistent with those determined for the same assets on a prospective basis in the distribution determination for that period.

Straight line depreciation of the RAB is determined in the RFM broadly as follows:

- The opening asset base as of 1 July 2010 by asset class is divided by the remaining life of each asset class.
- The actual capital expenditure for 2010-11 to 2013-14 inclusive, and the estimated capital expenditure for 2014-15 are divided by the economic or "standard" life of each asset class.

Similarly, straight line depreciation of the RAB is determined in the PTRM as follows:

- The opening asset base as of 1 July 2015 by asset class is divided by the remaining life of each asset class.
- The forecast capital expenditure for the regulatory control period 2015-20 inclusive is divided by the economic or "standard" life of each asset class.

A more detailed discussion on asset lives is set out in section 4.2.

There are some important considerations in determining the annual depreciation:

- Consistent with the AER's RFM, depreciation of capital expenditure is not recognised in the year in which the capital expenditure was incurred. Depreciation of capital expenditure in year t is recognised from year t+1 onwards.
- The "standard" economic lives for each asset class are those approved by the AER in its 2010-15 Distribution Determination.
- The remaining lives for each asset class for 2009-10 are those approved by the AER in its 2010-15 Distribution Determination. For each subsequent regulatory year, the remaining lives of each asset class reduced by 1 (unless the resulting remaining life is less than zero, in which case the remaining life is truncated to zero).

In accordance with the requirements of clause 6.5.5 of the NER, we can confirm that regulatory depreciation has been calculated:

- on the value of the assets in our RAB as at 1 July 2010 (in the RFM) and as at 1 July 2015 (in the PTRM) and for each subsequent regulatory year of the current and next regulatory control periods
- using a straight-line approach over the remaining lives of the asset classes for assets within the opening RAB. This reflects the nature of the assets or category of assets over their economic lives. It should be noted that within an asset class the individual assets may have a remaining life that is different to the weighted average remaining life calculated for the asset class
- using a straight line approach over the standard economic life applied to the actual capital expenditure (for 2010-11 to 2013-14 inclusive), estimated capital expenditure (for 2014-15) and forecast capital expenditure (for 2015-16 to 2019-20 inclusive). This reflects the nature of the assets or category of assets over their economic lives
- on the basis that the sum of the real value for any asset over its economic life (such real value being calculated as at the time the value of that asset was first included in the RAB) must be equivalent to the value at which that asset or category of assets was first included in the RAB.

The AER is not free to simply determine its own depreciation schedules. If the depreciation schedules proposed by Ergon Energy satisfy the requirements in clause 6.5.5(b) of the NER, they

must be used in calculating depreciation allowances, even if the AER prefers a different method for determining depreciation schedules.

In its Preliminary Determination, the AER assessed Ergon Energy's proposed depreciation schedules by comparing them with schedules produced using the AER's preferred method. The AER appears to take this approach on the basis that it considers depreciation schedules produced using its own method would satisfy the requirements of clause 6.5.5(b) of the NER.

This approach may be seen as complying with the requirements of clause 6.5.5(b) of the NER, *provided* the AER does not reject Ergon Energy's proposed depreciation schedules simply because they differ from those produced by the AER. It is still necessary for the AER to consider whether the schedules proposed by Ergon Energy satisfy the requirements of clause 6.5.5. If they do, the proposed schedules must be accepted, even if they differ from the AER's preferred approach.

In accordance with clause S6.1.3(12)(i) to (iv) of the NER, we confirm that our RFM and PTRM:

- categorise assets according to well accepted asset classes
- provide details of all amounts, values and inputs that we used to compile the depreciation schedules
- demonstrate that the depreciation schedules conform to the requirements set out in clause 6.5.5(b) of the NER.

4.2 Asset lives

4.2.1 Standard Lives

The standard lives for each asset class in the RFM and PTRM are the same as those approved by the AER for the regulatory control period 2010-15, with the exception of Type 5-6 meters, which have a standard life of 3 years. Type 5-6 meters was not a separate asset class in the regulatory control period 2010-15. In accordance with Schedule 1 clause 28.4 of the Reset RIN, a more detailed discussion of the standard life for Type 5-6 meters is provided in our supporting document 05.03.01 - (Revised) Default Metering Services Summary.

In accordance with Schedule 1 clause 28.3 of the Reset RIN, we do not propose any other changes to our standard lives for the existing asset classes in the RFM or PTRM for the regulatory control period 2015-20.

4.2.2 Remaining Lives

In its Preliminary Determination, the AER proposed to use a Weighted Average Remaining Life (WARL) approach for determining the remaining life of each asset class for the purposes of calculating the remaining lives of each asset class as at 1 July 2015. Whilst Ergon Energy broadly accepts this approach, we propose a modified WARL approach for calculating remaining asset lives in the regulatory control period 2015-20. The following sections outline our proposed approach.

Ergon Energy has adopted a similar approach for calculating remaining lives as adopted by Transgrid in its 2014-19 revised regulatory proposal. Specifically, Transgrid:

- proposed asset classes for:
 - Assets pre 2004-05
 - Assets 2004-09
 - Assets 2009-14
 - Assets 2014-19
- applied the AER's preferred WARL approach to the pre 2004-05, 2004-09 and 2009-14 asset classes for the purposes of calculating the weighted average remaining life as at 1 July 2014.

Providing asset classes categorised by regulatory period in this manner clearly delineates the assets added to the RAB in each regulatory control period since 2004-05. This makes the WARL calculation considerably more accurate, particularly for assets installed after 2004-05.

We note that in both the draft and final determinations, the AER accepted Transgrid's proposed asset classes and the WARL method to calculate the remaining asset lives as at 1 July 2014.

Ergon Energy has adopted a similar approach to that accepted by the AER in Transgrid's final determination. Specifically, Ergon Energy has:

- created asset classes for:
 - o assets installed pre 2009-10
 - o assets installed post 2009-10
- applied the AER's preferred WARL approach to these asset classes, modified such that the WARL extends to capital expenditure from 2009-10 to 2014-15 inclusive
- calculated the remaining lives for assets pre-2009-10 using the average depreciation approach
- used the AER's standard lives as set out in the Preliminary Determination for assets post-2009-10 in the WARL calculation.

Ergon Energy's approach for calculating the WARL is generally consistent with the AER's preferred approach set out in the Preliminary Determination. However, Ergon Energy's WARL calculations improve upon the AER's approach set out in the Preliminary Determination in the following ways:

- Separate asset classes have been created for asset pre and post 1 July 2009.
- The WARL is applied to capital expenditure from 2009-10 to 2014-15 inclusive, not 2010-11 to 2014-15 inclusive as proposed by the AER.

These differences, and why Ergon Energy considers these to be consistent with the NER, are described in more detail below.

Ergon Energy's proposed asset classes differ to those accepted by the AER in its Preliminary Determination

Whilst the assets grouped into each asset class have not changed, Ergon Energy has effectively split each of its existing asset classes into two, in order to cater for assets installed pre 2009-10 and post 2009-10. To distinguish between the pre and post 2009-10 asset classes, Ergon Energy has appended the label "2009-15" to each of the asset classes corresponding to asset installed post 1 July 2009.

This is consistent with the approach taken by Transgrid to group its assets by regulatory control period and we note that the AER has approved these asset classes. It is also consistent with the

requirements of clause 6.5.5.(b)(1) of the NER, which require that the depreciation schedules must depreciate using a profile that reflects the nature of the assets or category of assets over the economic life of that asset or category of assets. Splitting the asset classes by regulatory control period preserves the nature of the assets being depreciated (for example, assets in the Distribution Substations 2009-15 asset class are of the same type as those in the Distribution Substations asset class), whilst more accurately calculating the remaining life of the assets both pre and post 2009-10 over the economic life of the asset class.

Despite the change in asset classes between the regulatory control period 2010-15 and the regulatory control period 2015-20, there is no "disconnect" between regulatory control period. This is because we have not removed or changed any of the existing asset classes in the 2010-15 regulatory control period, and the total value of the assets classes pre 2009-10 and post 2009-10 can still be added together to give a single opening RAB value as of 1 July 2015.

The WARL is applied to capital expenditure from 2009-10 to 2014-15 inclusive, not 2010-11 to 2014-15 inclusive as proposed by the AER

Ergon Energy has applied the WARL to capital expenditure from 2009-10 to 2014-15 inclusive, rather than 2010-11 to 2014-15 inclusive as set out in the AER's Preliminary Determination. Ergon Energy has extended the AER's WARL approach to 2009-10 because:

- actual capital expenditure for 2009-10 by asset class is recorded in the RFM in the same way as actual capital expenditure from 2010-11 to 2014-15
- given that 2009-10 actual capital expenditure data is available, it would be inconsistent not to apply the WARL approach to capital expenditure from 2009-10 to 2014-15
- using 2009-10 actual capital expenditure removes the need to correct for the difference between forecast and actual capital expenditure in 2009-10 when calculating the 1 July 2015 remaining lives
- including 2009-10 gives an additional year of capital expenditure granularity in the WARL calculation, thereby improving the accuracy of the WARL, minimising the impact of averaging asset values and lives over time, and better reflecting the mix of depreciated assets in the RAB as at 1 July 2015.

Ergon Energy has modified the AER's RFM and PTRM in adopting this approach. We have:

- extended the number of asset classes accepted by the PTRM to 60 (and updated the depreciation and tax depreciation schedules to cater for the 60 asset classes)
- modified the WARL calculation to include 2009-10 capital expenditure.

These amended RFM and PTRM models are included in our revised Regulatory Proposal submission.

Houston Kemp Consulting was engaged to prepare an expert report which provides additional justification on the appropriateness of Ergon Energy's proposed remaining life calculation. Please refer to Houston Kemp Consulting's report *Analysis of Different Approaches to Calculating Remaining Lives* for further details.

The only exception to this approach is Type 5-6 meters, which have a remaining life of 5 years. A more detailed discussion of the remaining life for Type 5-6 meters is provided in our supporting document 05.03.01 - (Revised) Default Metering Services Summary.

The depreciation schedules proposed by Ergon Energy confirm with the requirements of clause 6.5.5(b) of the NER

In rejecting the method proposed by Ergon Energy for determining average remaining asset lives as at 1 July 2015, the AER found that:

- Ergon Energy's approach "consistently underestimates the remaining asset lives".
- Remaining asset lives determined using the AER's WARL approach "better reflect the nature of the assets over their economic lives".

The AER explained why it considered the WARL approach to be the better method for calculating the remaining lives of the assets in each asset class, stating:

"The most accurate way of estimating remaining asset lives is to track every asset individually. That is, record each asset added to the RAB and track its value over time."

However, the AER identifies several drawbacks with this approach, namely:

- because of the large number of assets to be added to the RAB over time, this approach places significant administrative costs on the business and regulatory, and
- not tracking assets individually may reduce volatility in revenues.

The AER recognised that, to reduce administrative costs, assets may be combined into classes, with an average remaining life assigned to the assets in each class. This average remaining life is re-calculated at each reset.

The AER assessed two methods by which average remaining lives could be calculated:

- a WARL approach (the approach favoured by the AER), and
- an average depreciation approach (the approach originally proposed by Ergon Energy).

The AER found:

"The remaining asset lives calculated by both the WARL and average depreciation approaches are not perfect compared with the approach of tracking assets individually. Some information is lost when assets are combined into a single asset class, and when new assets are added to that asset class. For this reason, we focus on the materiality of calculation distortions relative to the 'true' remaining asset lives (that is, remaining asset lives if assets were not aggregated into asset classes and they were not recalculated at each reset).

We prefer the WARL approach to the average depreciation approach because we consider it results in remaining asset lives that better reflect the economic life of the combined assets. It also results in depreciation schedules for the asset classes that reflect the nature of the assets over their economic lives."

The AER concluded that the WARL approach better resolves the problem created by combining old and new assets in a single class, producing a more balanced outcome in the long run.

Ergon Energy agrees with the AER that:

- the most accurate way of estimating remaining asset lives would be to track every asset individually; and
- an approach which produces average remaining lives (either WARL or a simple average) is imperfect compared with the more accurate approach of tracking assets individually.

Combining old and new assets in the same class must, by definition, result in an average life that differs from the actual remaining lives of the individual assets that constitute the class, irrespective of the averaging method used.

It follows that an approach which does not combine old and new assets in the same class (thereby avoiding the need to average those lives) must, by definition, produce depreciation schedules that reflect more accurately the remaining lives of the individual assets in each class. The findings of Houston Kemp provide further support for this conclusion.

As recommended by Houston Kemp, we have, for the purposes of this revised proposal, adopted the 'baseline' approach outlined above to determining remaining asset lives.

Under this approach, there is no grouping of pre and post 2009 assets in the same asset class. The distortion of remaining asset lives that results from combining old and new assets in the same class is avoided, and the resulting depreciation schedules align more closely with the actual remaining lives of individual assets than those produced using either the approach originally proposed by Ergon Energy in our October Regulatory Proposal or the WARL approach favoured by the AER.

Clause 6.5.5(a) of the NER requires depreciation to be calculated using the depreciation schedules nominated by Ergon Energy, provided those schedules conform with the requirements in clause 6.5.5(b). The AER rejected the approach first proposed by Ergon Energy because it resulted in a greater distortion of average remaining lives, relative to 'true' asset lives, than the AER's preferred approach and, as a consequence (and unlike the AER's preferred approach) did not adequately reflect the nature of the assets over their economic lives, as required by clause 6.5.5(b)(1) of the NER.

Ergon Energy's revised approach produces depreciation schedules that reflect more accurately, than either of the averaging approaches considered by the AER, the actual remaining asset lives of individual assets, since it significantly reduces:

- the extent to which old and new assets are combined in the same class
- the resulting distortion in average remaining lives, relative to the actual remaining lives of the assets in each class.

If the AER considers that:

- the most accurate way to depreciate assets is to assess their remaining lives individually
- its preferred WARL approach satisfies clause 6.5.5(b)(1) of the NER because, while imperfect, it produces average remaining lives that are sufficiently aligned to the remaining lives of the individual assets,

it must follow that Ergon Energy's revised approach also satisfies clause 6.5.5(b)(1) of the NER, since it significantly lessens the distortion of remaining asset lives when compared to the WARL approach.

The other findings made by the AER in its Preliminary Determination indicate that the depreciation schedules contained in this revised Regulatory Proposal otherwise satisfy the requirements of clause 6.5.5(b) of the NER.

Accordingly, clause 6.5.5(a)(2)(ii) of the NER requires that the revised depreciation schedules be used in calculating Ergon Energy's depreciation allowance. The NER do not permit the rejection of these revised depreciation schedules in favour of schedules that would be produced using a different method preferred by the AER, or a method which separates assets into classes covering different periods.

4.2.3 Tax Standard Lives

In accordance with Schedule 1 clause 29.4 of the Reset RIN, the tax standard lives for each asset class in the RFM and PTRM are the same as those approved by the AER for the regulatory control period 2010-15. An exception to this is Type 5-6 meters, which have a standard life of 25 years consistent with the Australian Taxation Office's tax Ruling TR 2014/4 – Effective Life Schedules.

We do not propose any other changes to our tax standard lives for the asset classes in the RFM or PTRM for the regulatory control period 2015-20.

4.2.4 Tax Remaining Lives

In its Preliminary Determination, the AER proposed to use a Weighted Average Remaining Life (WARL) approach for determining the tax remaining life of each asset class for the purposes of calculating the remaining lives of each asset class as at 1 July 2015. Whilst Ergon Energy broadly accepts this approach, we propose a modified WARL approach for calculating tax remaining asset lives in the 2015-20 regulatory control period.

The approach used for calculating the tax remaining lives for each asset class is the same as set out in section 4.2.2 above for the remaining lives. Please refer to this section for a detailed explanation of the approach used to calculate the remaining lives.

4.3 Indexation

The annual inflation for indexation of the RAB for the current regulatory control period applies both the AER's preferred methodology for calculating actual inflation and the Reserve Bank of Australia's (RBA) most recent Statement on Monetary Policy forecasts for 2013-14 and 2014-15.

Clause 6.5.1(e)(3) of the NER requires that:

The roll forward of the regulatory asset base from the immediately preceding regulatory control period to the beginning of the first regulatory year of a subsequent regulatory control period entails the value of the first mentioned regulatory asset base being adjusted for actual inflation, consistently with the method used for the indexation of the control mechanism (or control mechanisms) for Standard Control Services during the preceding regulatory control period.

We have applied inflation rates in the RFM based on the March to March indices for eight capital cities (from the Australian Bureau of Statistics) that were approved in the Initial and Annual Pricing Proposals for the regulatory control period 2010-15. Our RAB indexation for the current regulatory control period is set out in Table 14 below.

Table 14: Ergon Energy's RAB Indexation for 2010-15 (\$M nominal)

	2010-11	2011-12	2012-13	2013-14	2014-15
Opening RAB	7,148.95	7,843.82	8,375.96	9,034.88	9,649.23
Inflation Rate	3.33%	1.58%	2.50%	2.93%	2.75%
Indexation	238.06	123.93	209.40	264.72	265.35

Our proposed indexation for the next regulatory control period is based on the same approach as used to forecast the inflation rates for the current regulatory control period. That is, the method:

- first calculates the arithmetic average of the RBA's short-term Consumer Price Index (CPI) inflation forecasts published in Table 6.1 of its August 2014 Statement on Monetary Policy4 for 2013-14 to 2015-16 inclusive
- then calculates the geometric mean of the CPI values over the 10-year period 2013-14 to 2022-23 using the calculated CPI values for 2013-14 to 2015-16 (as described above) and the midpoint of the RBA's target CPI inflation range (2.5%) for each of the years 2016-17 to 2022-23 inclusive. This yields forecast CPI inflation values of 2.55% for each year of the regulatory control period 2015-20.

Our proposed RAB indexation for the regulatory control period 2015-20 is set out in Table 15 below.

	2015-16	2016-17	2017-18	2018-19	2019-20
Opening RAB	10,055.83	10,674.58	11,225.76	11,750.05	12,252.24
Inflation Rate	2.55%	2.55%	2.55%	2.55%	2.55%
Indexation	256.42	272.20	286.26	299.63	312.43

Table 15: Ergon Energy's RAB Indexation for 2015-20 (\$M nominal)

⁴ <u>http://www.rba.gov.au/publications/smp/2014/aug/html</u>, p71

5 Operating expenditure

For the purposes of the building block calculation we have included an allowance for operating expenditure for each year of the regulatory control period 2015-20. Operating expenditure consists of:

- Network Operating Costs
- Total Network Maintenance Costs
- Total Other Operating and Maintenance (O&M) Costs
- Debt raising costs

The table below sets out the forecast operating expenditure for each regulatory year of the regulatory control period 2015-20. These values are included as inputs to the PTRM.

Table 16: Ergon Energy's forecast operating expenditure 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
Network Operating Costs	34.99	37.41	39.80	41.76	43.89
<i>plus</i> Total Network Maintenance Costs	244.33	261.27	278.07	291.91	306.87
plus Total Other O&M Costs	63.22	65.82	68.42	70.99	73.78
plus Debt raising costs	12.19	12.94	13.61	14.24	14.85
Total Operating expenditure	354.73	377.44	399.89	418.91	439.39

5.1 Network Operating and Maintenance costs and Other O&M costs

Network Operating and Maintenance costs, as well as Other O&M costs are discussed in more detail in our supporting document 06.01.01 – (Revised) Operating Forecast Expenditure Summary Document.

5.2 Debt raising costs

Debt raising costs are calculated in accordance with the AER's PTRM in accordance with the following approach for each year of the regulatory control period 2015-20:

- The inflated nominal opening RAB for 2015-16 is multiplied by the product of the debt raising cost benchmark (0.197%) and the proportion of debt funding (60%). This gives the nominal debt raising cost for the regulatory year.
- The nominal debt raising cost is divided by the cumulative CPI escalator for the regulatory year to de-escalate to 2014-15 dollars. This gives the real debt raising cost for the regulatory year expressed in 2014-15 dollars.

5.3 Revenue Adjustments

Revenue Adjustments amounts for each regulatory year include:

- EBSS adjustments. These are covered in more detail in section 7.1.1 of this document and in our supporting document 03.01.03 (Revised) Application of Incentives Schemes
- DMIS adjustments. These are covered in more detail in section 7.1.3 of this document and in our supporting document 03.01.03 (Revised) Application of Incentives Schemes
- DUOS unders and overs in 2015-16. This is covered in more detail in section 7.2.1 of this document and in our supporting document 03.01.02 (Revised) Other Revenue Adjustments.
- Shared asset decrements. These are covered in more detail in section 7.3 of this document and in our supporting document 03.01.02 (Revised) Other Revenue Adjustments.

6 Corporate income tax

As required by Schedule 1 clause 29.1 and clause 29.4 of the Reset RIN, we have calculated the estimated cost of corporate income tax (and the associated tax asset base roll forward required to determine the annual tax depreciation) for each year of the regulatory control period 2015-20, by applying the AER's Version 03 PTRM without further modification. The PTRM calculates the estimated cost of corporate income tax in a manner consistent with clause 6.5.3 of the NER, which requires that the estimated cost of corporate income tax for each regulatory year (ETC_t) be estimated as follows:

$$ETC_t = (ETI_t \times r_t) (1 - \gamma)$$

where:

 ETI_t is the estimated taxable income for regulatory year 't' earned by a benchmark efficient entity through the provision of Standard Control Services if such an entity, rather than Ergon Energy, operated our business. These values are generated by the AER's PTRM, which determines the taxable income as follows:

Taxable income = ARR (unsmoothed) + annual capital contributions – annual operating expenditure – tax depreciation – interest (i.e. return on debt)

- r_t is the expected statutory income tax rate for that regulatory year as determined by the AER, and γ
- γ is the value of imputation credits (gamma), which is itself the product of the Distribution Rate (or Payout Rate) and the utilisation rate (theta).

We are proposing a gamma of 0.25, which reflects a distribution rate of 0.7 and theta of 0.35. This differs from the AER's Rate of Return Guideline, which proposes values for both these parameters of 0.7. We do not consider that 0.7 is the best value for theta, having regard to the requirements of the NER for the following reasons.

As noted in Appendix C of the revised Regulatory Proposal, SFG's Gamma Report identifies a number of issues with the approach taken by the AER in developing its Rate of Return Guideline. It conducts a detailed review of the AER's conceptual interpretation of theta and highlights some fundamental flaws. SFG clearly demonstrates that the relevant task is to establish a market-based value of theta. This also invalidates the equity ownership, tax statistics and 'conceptual goalposts' approach that have been referred to by the AER.

We concur with this view. The gamma parameter is intended to reflect the value that investors place on franking credits in establishing the rate of return they require from the efficient benchmark firm. This has to be a market value. The AER's conclusion that this should only reflect the extent to which imputation credits might be used to reduce personal tax is erroneous and can (and has) resulted in gamma being overestimated. If the value that investors are assumed to derive from imputation credits is overstated, this will mean that their required rate of return will be underestimated.

SFG has also undertaken an updated empirical analysis of theta using dividend drop-off studies and other market value studies. This analysis concludes that:

 0.35 remains the best estimate of theta at the current time using a dividend drop-off approach (based on the SFG approach, which has been subject to unprecedented scrutiny)
• other market value studies support an estimate between zero and 0.35.

The SFG analysis supports the conclusion that a theta of 0.35 is more likely to be at the upper bound of a reasonable range.

Our estimated cost of corporate income tax for each year of the regulatory control period 2015-20 is set out in the table below.

Table 17: Ergon Energy's estimated cost of corporate income tax for Standard Control Services for 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
Corporate Income Tax	128.22	158.74	168.00	176.59	170.17
<i>less</i> Value of Imputation Credits	(32.05)	(39.69)	(42.00)	(44.15)	(42.54)
Estimate cost of corporate income tax	96.16	119.06	126.00	132.45	127.63

Further details on the corporate income tax calculation are provided in Appendix C of the revised Regulatory Proposal.

6.1 Departures

In accordance with Schedule 1 clause 29.3 of the Reset RIN, we must provide details of each departure from the AER's PTRM for the estimated cost of corporate income tax and the reasons for that departure. In calculating the estimated cost of corporate income tax for the regulatory control period 2015-20, we have adopted the AER's Version 03 PTRM.

We note, however, that we have departed from the AER's Rate of Return Guideline for the setting of gamma (as highlighted earlier in this chapter).

6.1.1 Recovery of tax payable on contributed and gifted assets received from large customer connections in 2015-20

In the regulatory control period 2010-15, we note that the tax paid as a result of the contributed and gifted assets we received from large customer connections was not able to be recovered from either the Standard Control Service revenue requirement or any of the Alternative Control Service pricing mechanisms approved by the AER in the 2010-15 Distribution Determination. We will not be recovering these foregone costs in the regulatory control period 2015-20, however in our Regulatory Proposal we proposed to recover the forecast tax payable as a result of the contributed and gifted assets we expect to receive from large customer connections in the regulatory control period 2015-20.

Our analysis of the AER's Preliminary Determination models indicates that the AER has removed from our proposed PTRM all gifted and contributed assets associated with Large Customer Connections in the regulatory control period 2015-20. There is no explanation of their reasons for this and we assume this an oversight by the AER. The inclusion of these values does not impact the value of the RAB for Standard Control Services (reflecting the prepayment, contribution of

gifting). However, the omission of the values from the PTRM means that the tax allowance is understated.

In our revised Regulatory Proposal we have adopted the same approach as set out in our Regulatory Proposal for recovering the tax payable on contributed and gifted assets we will receive from large customer connections in the regulatory control period 2015-20:

- The forecast contributed and gifted asset values from large customer connections for each regulatory year of regulatory control period 2015-20 were developed by asset class. Please refer to our supporting document 07.00.03 – (Revised) Forecast Expenditure Summary Customer Initiated Capital Works 2015 to 2020 for further details on how these forecasts were developed.
- 2. The forecast contributed and gifted asset values from large customer connections, by asset class, for each regulatory year of the regulatory control period 2015-20 were then added to the Standard Control Services capital expenditure forecasts by asset class for each regulatory year of the regulatory control period 2015-20. These combined values were then entered into the PTRM for Standard Control Services
- 3. The forecast contributed and gifted asset values from large customer connections, by asset class, for each regulatory year of the regulatory control period 2015-20 were also added to the Standard Control Services capital contribution forecasts by asset class for each regulatory year of the regulatory control period 2015-20. These combined values were also entered into the PTRM for Standard Control Services.

By virtue of the operation of the Standard Control Services PTRM, this method ensures that we receive the tax allowance we require to recover the tax payable for contributed and gifted assets from large customer connections during the regulatory control period 2015-20. This is because row 39 of the Analysis tab in the PTRM includes the capital contributions as additional tax income, and row 43 of the Analysis tab in the PTRM uses the value of capital contributions in determining the tax depreciation for each regulatory year. In this way, the contributed and assets from large customer connections during the regulatory control period 2015-20 are included in the tax expense and tax payable calculations in the PTRM (and hence the ARR for Standard Control Services).

For this reason, the tax payable (and hence the tax allowance) in the regulatory control period 2015-20 will be higher than in the previous regulatory control period.

Forecast contributed and gifted asset values from large customer connections for each regulatory year of regulatory control period 2015-20 are not inappropriately recovered twice from revenues for Standard Control Services. By adding the contributed and gifted assets from large customer connections during the regulatory control period 2015-20 to both the forecast capital expenditure and the forecast capital contributions for each year of the regulatory control period 2015-20, the PTRM removes the contributed and gifted assets from large customer connections in calculating the net capital expenditure in rows 263 to 322 of the Input tab. Only the Standard Control Services net capital expenditure is used to calculate the return on and depreciation allowance for Standard Control Services.

7 Revenue increments/decrements

This chapter sets out the revenue increments and decrements that form part of our building block revenue requirement for Standard Control Services. Clause 6.4.3(a) of the NER outlines the following revenue increments and decrements:

(5) the revenue increments or decrements (if any) for that year arising from the application of any efficiency benefit sharing scheme, capital expenditure sharing scheme, service target performance incentive scheme, demand management and embedded generation connection incentive scheme or small-scale incentive scheme – see subparagraph (b)(5);

(6) the other revenue increments or decrements (if any) for that year arising from the application of a control mechanism in the previous regulatory control period – see paragraph (b)(6);

(6A) the revenue decrements (if any) for that year arising from the use of assets that provide standard control services to provide certain other services – see subparagraph (b)(6A).

The table below summarises the ARR impacts during the regulatory control period 2015-20 as a result of these revenue increments and decrements contemplated by clause 6.4.3(a)(5).

\$m (nominal)	2015-16	2016-17	2017-18	2018-19	2019-20
EBSS (2010-15)	34.61	50.42	68.83	(20.25)	-
DMIS (Part A, DMIA) 2015-20	1.03	1.05	1.08	1.11	1.13
STPIS (2010-15)	22.75	(7.56)	-	-	-
DMIS carryover (2010-15)	-	(1.90)	-	-	-

Table 18: Ergon Energy's revenue increments and decrements included in the ARR for 2015-20

The EBSS and DMIS (Part A, Demand Management Innovation Allowance (DMIA) revenue increments are included in the unsmoothed building block revenue calculation in the PTRM. The STPIS and DMIS carryover amounts are not included in the PTRM as adjustments are made to the ARR during the annual pricing proposal process.

The table below summarises the ARR impacts during the regulatory control period 2015-20 as a result of the revenue increments and decrements contemplated by clause 6.4.3(a)(6) of the NER.

Table 19: Ergon Energy's other building block revenue increments and decrements for 2015-20

\$m (nominal)	2015-16	2016-17	2017-18	2018-19	2019-20
Included in building block calculation					
Residual DUOS unders and overs account balance as at 30 June 2015	58.56	-	-	-	-

\$m (nominal)	2015-16	2016-17	2017-18	2018-19	2019-20				
Recognised in annual pricing proposal after building block revenue calculation									
2013-14 DUOS under-recovery	67.10	-	-	-	-				
Transitional capital contribution unders and overs (2010-15)	85.12	114.66	-	-	-				
Transitional shared assets unders and overs (2010-15)	(3.58)	(3.48)	-	-	-				
Cost pass throughs - FIT	135.03	125.48	-	-	-				

Refer to our supporting document 04.01.00 – (*Revised*) Compliance with Control Mechanisms for further details on the formulaic expression of the Total Annual Revenue and how these adjustments are recognised.

The table below summarises the ARR impacts during the regulatory control period 2015-20 as a result of the shared asset revenue decrement contemplated by clause 6.4.3(a)(6A) of the NER.

Table 20: Ergon Energy's estimated revenue adjustment associated with the use of shared assets in 2015-20

\$m (nominal)	2015-16	2016-17	2017-18	2018-19	2019-20
Revenue adjustment - shared assets 2015-20	(6.71)	(6.89)	(7.06)	(7.24)	(7.43)

The shared asset revenue decrements are included in the unsmoothed building block revenue calculation in the PTRM.

The remainder of this chapter outlines each of the forecast adjustments set out in the tables above in more detail.

7.1 Incentive schemes

The AER's Framework and Approach Paper proposed to apply the following incentive schemes to us in the regulatory control period 2015-20. The objective of this is to provide financial incentives to DNSPs to make efficient investment decisions and to maintain the efficiency of their expenditure, performance and services over time:

- **DMIS** provides incentives to electricity distribution businesses to commission efficient nonnetwork solutions, such as distributed generation, to meet network constraints
- EBSS rewards electricity distribution businesses for efficiency gains and penalises them for efficiency losses as benchmarked against the AER's operating expenditure forecasts. Any gains and losses outstanding at the end of a regulatory control period are carried over into the next. We note that whilst the AER's Framework and Approach Paper proposed to apply the EBSS in the regulatory control period 2015-20, the AER in its Preliminary Determination decided not to apply the EBSS. Ergon Energy does not agree with the AER's decision, and instead proposes to apply the EBSS in the regulatory control period 2015-20 in accordance with the Framework and Approach and consistent with Ergon Energy's Regulatory Proposal

- **STPIS** encourages electricity distribution businesses to maintain and improve service performance by delivering financial rewards for out-performance, or by imposing financial penalties for under-performance against historical service standard data
- CESS rewards electricity distribution businesses for underspends and penalises them for overspends as benchmarked against their capital expenditure program approved by the AER for the regulatory control period. The AER is able to require gains and losses outstanding at the end of a regulatory control period to be carried over into the next. When an electricity distribution business overspends its capital allowance, the CESS allows the AER to undertake an ex-post review of capital works and adjust the Standard Control Service RAB for imprudent or inefficient capital costs.
- **SSIS** the AER can apply an incentive scheme to an electricity distribution business as part of a regulatory determination. Unlike the other incentive schemes it would apply to only that business.

The remainder of this section sets out at a high level:

- how each incentive scheme was applied during the regulatory control period 2010-15 (where applicable)
- how each incentive scheme will be applied during the regulatory control period 2015-20 (where applicable)
- the forecast annual revenue impact of the schemes on the ARR for the regulatory control period 2015-20.

The remainder of this section therefore sets out the outworking of the regulatory control period 2010-15 incentive schemes. A more detailed explanation of the calculations supporting these incentive scheme adjustment values is provided in 03.01.03 – (*Revised*) Application of Incentive Schemes.

7.1.1 Efficiency Benefit Sharing Scheme

The EBSS seeks to provide a financial incentive for us to improve the efficiency of our operating expenditure and to share any resulting efficiency gains (or losses) with customers. Any efficiency gains (or losses) are retained by us for five years after the gain (or loss) is realised. This means the EBSS revenue adjustment in the next regulatory control period relates to performance under the EBSS in the current regulatory control period.

In the first two to three years of the current regulatory control period we overspent our operating allowance for EBSS but have turned this around in the third/fourth year and we are estimating a further reduction in operating costs in the fifth year.

EBSS applied in current regulatory control period

The EBSS applied in the regulatory control period 2010-15 will impact the ARR values in the regulatory control period 2015-20. These revenue increments form part of the revenue adjustments in the PTRM and are ultimately included in the unsmoothed revenue calculation. The table below sets out the annual revenue adjustments in the regulatory control period 2015-20 as a result of the operation of the EBSS during the regulatory control period 2010-15.

Table 21: Ergon Energy's forecast revenue adjustments associated with EBSS, 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
EBSS Adjustments	34.61	50.42	68.83	(20.25)	-

The operating expenditure outcomes that give rise to the ARR adjustments are set out in the table above, and are explained in further detail in our supporting document 03.01.03 – (Revised) Application of Incentive Schemes.

EBSS for the next regulatory control period

We do not accept the AER's decision not to apply the EBSS in the regulatory control period 2015-20. However, we note that applying the EBSS in the regulatory control period 2015-20 will impact the ARR in the regulatory control period 2020-25 only, and hence will have no impact on the regulatory control period 2015-20.

We are not proposing any changes to the operation or application of the EBSS in the regulatory control period 2015-20 from that set out by the AER in its Framework and Approach and that proposed in our October Regulatory Proposal.

7.1.2 Service Target Performance Incentive Scheme (STPIS)

The STPIS rewards us when we improve average service to customers and penalises us for a reduction in average service to customers. The rewards or penalties are applied by adjusting the amount of allowed revenue in a year in accordance with the mechanism set out in the distribution determination and the relevant STPIS guidelines. We currently have a maximum reward or penalty of \pm 2% of our allowed annual revenues in the regulatory control period 2010-15 and this will remain at \pm 2% in the regulatory control period 2015-20.

STPIS applied in current regulatory control period

As set out in our supporting document 03.01.03 – (*Revised*) Application of Incentive Schemes, the following table summarises the forecast revenue adjustments likely to occur as a result of the operation of the STPIS in 2013-14. These revenue increments are not included in the PTRM as adjustments are made to the ARR during the annual pricing proposal process.

	2015-16	2016-17	2017-18	2018-19	2019-20
STPIS revenue adjustments from regulatory control period 2010-15	22.75	(7.56)	-	-	-

Table 22: Ergon Energy's forecast revenue adjustments associated with STPIS, 2015-20 (\$M nominal)

STPIS for the next regulatory control period

Our supporting document 03.02.02 – (*Revised*) Proposed Application of STPIS 2015-20 for the regulatory control period 2015-20 sets out our proposed STPIS targets for the regulatory control period 2015-20.

We have not forecast revenue adjustments due to the operation of the STPIS within the regulatory control period 2015-20. This is because we plan and operate our network in such a way as to meet the STPIS targets each year and do not specifically target higher or lower performance relative to these STPIS targets. Meeting the STPIS targets would result in no revenue adjustments under the operation of the STPIS and for this reason, no revenue adjustments due to the STPIS in the regulatory control period 2015-20 have been forecast.

7.1.3 Demand Management Incentive Scheme

The DMIS seeks to provide incentives to us to implement efficient non-network alternatives for managing expected demand on the network and efficiently connect embedded generators. In its Framework and Approach Paper, the AER proposed to apply Part A of the DMIS in the next regulatory control period (i.e. the DMIA). Consistent with the Framework and Approach Paper, we have proposed a total DMIA allowance of \$5 million (in real 2014-15 dollars) over the regulatory control period 2015-20.

7.1.3.1 DMIS applied in current regulatory control period

As set out in our supporting document 03.01.03 – (*Revised*) Application of Incentive Schemes, the following table summarises the regulatory control period 2010-15 actual and estimated DMIA expenditure:

 Table 23: Ergon Energy's actual and estimated revenue adjustments associated with DMIS, 2010-15 (nominal dollars)

	2010-11	2011-12	2012-13	2013-14	2014-15
DMIS (Part A, DMIA)	0.50	0.58	0.93	0.87	1.00

7.1.3.2 DMIS Carryover from the current regulatory control period

In accordance with the Total Annual Revenue formulas set out in our supporting documents 04.01.00 – (Revised) Compliance with Control Mechanisms and 04.01.05 – (Revised) Control Mechanism Model, the DMIS carryover amount from the regulatory control period 2010-15 is presented in the table below. These revenue decrements are not included in the PTRM as adjustments are made to the ARR during the annual pricing proposal process.

Table 24: Ergon Energy's estimated revenue decrements from the DMIS carryover, 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
DMIS carryover (2010-15)	-	(1.90)	-	-	-

Refer to our supporting documents 04.01.00 – (*Revised*) Compliance with Control Mechanisms and 04.01.05 – (*Revised*) Control Mechanism Model for further details on the calculation and application of the DMIS Carryover from the regulatory control period 2010-15.

No further adjustments to the Total Annual Revenue for this DMIS Carryover will occur during the regulatory control period 2015-20.

7.1.3.3 DMIS for the next regulatory control period 2015-20

The AER noted in its Framework and Approach Paper that it may develop and implement a new DMIS during the next regulatory control period, depending on the progress of the Australian Energy Market Commission Power of Choice rule change process. For these reasons, the AER has proposed to allow a \$5 million DMIA (\$1 million in real 2014-15 dollars each year), consistent with our scheme allowance under the current regulatory control period and consistent with the AER's Preliminary Determination. For revenue modelling purposes, Ergon Energy has included the \$5 million DMIA as a revenue adjustment of \$1 million per annum in 2014-15 dollars. To avoid double counting of the allowance, the DMIS has been removed from Ergon Energy's proposed base year operating expenditure and hence is no longer included in our proposed operating expenditure for the regulatory control period 2015-20.

The following table summarises the revenue adjustments included in the ARR for the DMIS for the regulatory control period 2015-20:

Table 25: Ergon Energy's forecast revenue adjustments associated with DMIS, 2015-20 (real 2014-15 \$M)

	2015-16	2016-17	2017-18	2018-19	2019-20
DMIS (Part A, DMIA)	1.03	1.05	1.08	1.11	1.13

Further detail is provided in our supporting document 03.01.03 – (*Revised*) Application of Incentive Schemes.

7.1.4 Capital Expenditure Sharing Scheme

We note that the CESS will not apply for the regulatory control period 2015-20 and as such it has not been incorporated in the development of the proposed ARR building blocks for the regulatory control period 2015-20.

7.1.5 Small Scale Incentive Scheme

We note that the SSIS will not apply for the regulatory control period 2015-20 and as such it has not been incorporated in the development of the proposed ARR building blocks for the regulatory control period 2015-20.

7.2 Adjustments from the application of the control mechanisms in the current regulatory control period

This section sets out the forecast revenue increments and decrements arising from the application of the control mechanism in place for the regulatory control period 2010-15, in accordance with clause 6.4.3(a)(6) of the NER.

7.2.1 Clearing the 2013-14 DUOS under-recovery

As noted in our supporting document 04.01.00 - (Revised) Compliance with Control Mechanisms, we propose to clear the entire 2013-14 DUOS under-recovery as a \$67.1 million adjustment in the Total Annual Revenue calculation, as set out in the table below.

Table 26: Ergon Energy's estimated revenue increments for clearing the 2013-14 under-recovery (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
2013-14 DUOS Under-recovery adjustment	67.10	-	-	-	-

Note that this adjustment is not a carryover adjustment made in the PTRM. Further detail is provided in our supporting document 04.01.00 – (*Revised*) Compliance with Control Mechanisms.

7.2.2 Clearing the residual DUOS unders and overs account balance

Consistent with the approved 2014-15 Pricing Proposal, we propose to clear the residual balance left in our DUOS unders and overs account as at 30 June 2015 as a carry forward adjustment in the PTRM. Specifically, the carry forward amount included in the PTRM includes a value of \$58.56 million (nominal) in 2015-16 as set out in the table below.

Table 27: Ergon Energy's estimated revenue increments for clearing unders and overs, 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
DUOS Unders and Overs (2010-15)	58.56	-	-	-	-

Further detail is provided in our supporting documents 03.01.02 – (Revised) Other Revenue Adjustments, 04.01.00 – (Revised) Compliance with Control Mechanisms and 04.01.05 – (Revised) Control Mechanism Model.

7.2.3 DUOS unders and overs in the regulatory control period 2015-20

We note that the adjustments described above only account for the unders and overs balance that has accumulated to date in the regulatory control period 2010-15. Adjustments for any unders and overs arising from the application of the control mechanism in the regulatory control period 2015-20 will be managed as per the proposed approach set out in Chapter 4 of the revised Regulatory Proposal. These revenue increments and decrements do not form part of the Carryover Amounts in the PTRM and are ultimately excluded in the unsmoothed revenue calculation.

7.2.4 Transitional capital contribution unders and overs

In accordance with the Total Annual Revenue formula set out in supporting documents 04.01.00 – (*Revised*) Compliance with Control Mechanisms and 04.01.05 – (*Revised*) Control Mechanism Model, the actual under or over-recovery of capital contributions revenue (compared to forecast in AER's 2010-15 Distribution Determination) is presented in the table below. These revenue

increments are not included in the PTRM as adjustments are made to the ARR during the annual pricing proposal process.

 Table 28: Ergon Energy's estimated revenue increments for clearing transitional capital contribution unders and overs, 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
Transitional capital contribution unders and overs (2010-15)	85.12	114.66	-	-	-

Refer to our supporting documents 04.01.00 - (Revised) Compliance with Control Mechanisms and 04.01.05 - (Revised) Control Mechanism Model for further details on the calculation and application of the transitional capital contribution unders and overs from the regulatory control period 2010-15 and the regulatory control period 2015-20.

As this adjustment is as a result of a transitional arrangement from the regulatory control period 2010-15, no further adjustments to the Total Annual Revenue for the clearing of transitional capital contribution unders and overs will occur during the regulatory control period 2015-20.

7.2.5 Transitional shared assets unders and overs

In accordance with the Total Annual Revenue formula set out in our supporting documents 04.01.00 – (Revised) Compliance with Control Mechanisms and 04.01.05 – (Revised) Control Mechanism Model, the actual under or over-recovery of shared asset revenue (compared to forecast in AER's 2010-15 Distribution Determination) is presented in the table below. These revenue decrements are not included in the PTRM as adjustments are made to the ARR during the annual pricing proposal process.

Table 29: Ergon Energy's estimated revenue decrements for clearing transitional shared assets unders and overs, 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
Transitional shared assets unders and overs (2010-15)	(3.58)	(3.48)	-	-	-

Refer to our supporting documents 04.01.00 - (Revised) Compliance with Control Mechanisms and 04.01.05 - (Revised) Control Mechanism Model for further details on the calculation and application of the transitional shared assets unders and overs from the regulatory control period 2010-15 and the regulatory control period 2015-20.

As this adjustment is as a result of a transitional arrangement from the regulatory control period 2010-15, no further adjustments to the Total Annual Revenue for the clearing of transitional shared assets unders and overs will occur during the regulatory control period 2015-20.

7.2.6 Cost Pass-Through - Feed-in tariffs

In accordance with the Total Annual Revenue formulas set out in our supporting documents 04.01.00 – (Revised) Compliance with Control Mechanisms and 04.01.05 – (Revised) Control Mechanism Model, the estimated cost pass-through amounts as a result of the Solar Bonus

Scheme feed-in tariff payments that occurred during the regulatory control period 2010-15 are presented in the table below. These revenue increments are not included in the PTRM as adjustments are made to the ARR during the annual pricing proposal process.

Table 30: Ergon Energy's estimated revenue increments for cost pass throughs- Feed-in tariffs, 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
Cost Pass-throughs – Feed-in tariffs	135.03	125.48	-	-	-

Refer to our supporting documents 04.01.00 - (Revised) Compliance with Control Mechanisms and 04.01.05 - (Revised) Control Mechanism Model for further details on the calculation and application of the feed-in tariff cost pass-throughs from the regulatory control period 2010-15.

No further adjustments to the Total Annual Revenue for feed-in tariff cost pass-throughs will occur during the regulatory control period 2015-20. Refer to our supporting document *04.01.02 – (Revised) Jurisdictional Schemes* for details on the recovery of feed-in tariff costs in the regulatory control period 2015-20.

7.3 Shared assets

In its Preliminary Determination, the AER rejected our proposal to include the 2012-13 adjustment for assets used to provide Alternative Control Services. Ergon Energy does not accept this decision. We have updated our adjustment for assets used to provide Alternative Control Services to reflect 2013-14 information.

We are of the view that including the 2013-14 adjustment for assets used to provide Alternative Control Services aligns with the principles of the shared asset mechanism outlined in the AER's Shared Asset Guideline, that customers should not pay for more than their fair share for shared assets and that service providers may propose their own cost reductions. Further, the proposed revenue adjustment is equivalent to the control, which sets a cap on the quantum of the cost reduction.

We note that the Shared Asset Guideline only contemplates the situation where assets are used to provide Standard Control Services and unregulated services. The Shared Asset Guideline does not appear to consider the situation where assets are used to provide Standard Control Services and Alternative Control Services. Given this, we propose to continue to adjust for Alternative Control Services in our revenue adjustment calculations.

Consistent with our Regulatory Proposal submission and as noted in 03.01.02 – (Revised) Other Revenue Adjustments, the proposed revenue adjustment for the regulatory control period 2015-20 is sum of:

- the control for 2013-14 calculated in accordance with the AER's Shared Assets Guideline for the assets used to provide Unregulated Services.
- the 2013-14 adjustment calculated by us for the assets used to provide Alternative Control Services.

The resulting sum is indexed using the forecast inflation rate. These revenue increments form inputs to the PTRM and are ultimately included in the unsmoothed building block revenue

calculation. The following table summarises the proposed revenue adjustments (decrements) resulting from the use of shared assets for the regulatory control period 2015-20.

Table 31: Ergon Energy's adjustment for shared assets for 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
Total Revenue Adjustment	(6.71)	(6.89)	(7.06)	(7.24)	(7.43)

Further detail is provided in our supporting document 03.01.02 – (*Revised*) Other Revenue Adjustments.

8 Annual revenue requirement

The table below provides a summary of the total (unsmoothed) ARR, the ARR building blocks and the X factors / smoothed ARR for each year of the regulatory control period 2015-20.

	2015-16	2016-17	2017-18	2018-19	2019-20
Return on capital	744.94	790.77	831.60	870.44	907.65
plus Regulatory Depreciation	162.28	179.16	168.04	171.13	148.53
plus Operating expenditure	354.73	377.44	399.89	418.91	439.39
plus Corporate income tax	96.16	119.06	126.00	132.45	127.63
plus Revenue adjustments	87.48	44.58	62.84	(26.39)	(6.29)
Building Block Revenue (unsmoothed)	1,445.58	1,511.01	1,588.38	1,566.54	1,616.90
ARR (smoothed)	1,137.71	1,522.33	1,709.11	1,712.72	1,716.33
X Factors	36.63%	(30.48%)	(9.48%)	2.28%	2.28%

Table 32: Ergon Energy's ARR for 2015-20 (\$M nominal)

The values set out in the table above for 2015-16 are consistent with those determined by the AER in its Preliminary Determination. With the exception of the X factors for 2016-17 through to 2019-20 inclusive, the values set out in the table were calculated using the AER's PTRM. Therefore, we do not consider these calculations to be a departure from the AER's underlying methods and calculations for determining the unsmoothed and smoothed ARR values.

8.1 Comparison to forecast annual revenue requirement for regulatory control period 2010-15

We have examined the differences between the forecast ARR building blocks from the regulatory control period 2010-15 and the regulatory control period 2015-20. In order to do so, the forecast ARR building blocks from the regulatory control period 2010-15 were escalated to 2014-15 dollars and the forecast ARR building blocks from the regulatory control period 2015-20 were de-escalated to 2014-15 dollars.

The table below sets out the total ARR for the regulatory control periods 2010-15 and 2015-20 (in real 2014-15 dollars to allow comparison), and the differences in each of the forecast ARR building blocks between the regulatory control periods 2010-15 and 2015-20.

Table 33: Difference in Standard Control Services building blocks between regulatory control period 2010-15 and 2015-20 (real 2014-15 \$M)

Building Block	Differences between regulatory control periods	Total Revenue
Total ARR Regulatory Control Period 2010-15		7,113.17
Difference in Return on capital	(740.57)	
Difference in Regulatory Depreciation	(19.58)	
Difference in Operating expenditure	(211.69)	
Difference in Corporate income tax	206.96	
Difference in Revenue adjustments	813.02	
Total ARR Regulatory Control Period 2015-20		7.161.31

The following waterfall chart depicts these differences graphically.



The changes associated with each Standard Control Service building block are briefly explained below:

• The reduction in the return on asset is largely due to the lower nominal vanilla WACC of 7.41% (compared to the 9.72% nominal vanilla WACC in the regulatory control period

2010-15). This also represents a greater reduction in the return on asset than was originally proposed in the October Regulatory Proposal.

- The reduction in operating expenditure is due to lower operating expenditure forecasts for the regulatory control period 2015-20, compared to that forecast for the regulatory control period 2010-15. This also represents a greater reduction in operating expenditure than what was originally proposed in the October Regulatory Proposal.
- The decrease in the regulatory depreciation is due to the impact of using the WARL (rather than the average depreciation approach) for calculating remaining lives in the regulatory control period 2015-20. We note that in the October Regulatory Proposal, the regulatory depreciation was anticipated to be higher in the regulatory control period 2015-20 than in the regulatory control period 2010-15.
- The increase in the corporate income tax is largely due to:
 - the lower operating expenditure and interest expense (the return on debt rate is lower) in the regulatory control period 2015-20
 - o higher capital contributions in the regulatory control period 2015-20
 - o no tax loss carried forward in the regulatory control period 2015-20.

We note that the proposed corporate income tax is lower than that proposed in the October Regulatory Proposal.

- The difference in total revenue adjustments consists of:
 - the cessation of the transitional arrangement for the treatment of capital contributions as described in section 2.1.2.2. As of 1 July 2015, we will no longer retain the value of contributed assets in the RAB and therefore will no longer deduct the value of capital contributions from the ARR upfront. The majority of the increase in revenue adjustments is due to the apparent uplift associated with the removal of the upfront revenue deduction
 - higher shared assets adjustments
 - higher carryover amounts from the regulatory control period 2010-15 (including DUOS unders and overs).

8.2 Departures

Other than the changes to the AER's WARL calculation and the inclusion of asset classes for pre 2009-10 and post 2009-10 (as set out in section 4.2.2), Ergon Energy is not proposing any departures from the building block proposal requirements set out in Part C of the NER.

9 X factors

9.1 Relevant NER requirements

Clause 6.5.9 of the NER requires that:

(a) A building block determination is to include the X factor for each control mechanism for each regulatory year of the regulatory control period.

- (b) The X factor:
 - (1) Must be set by the AER with regard to our total revenue requirement for the regulatory control period
 - (2) Must be such as to minimise, as far as reasonably possible, variance between expected revenue for the last regulatory year of the regulatory control period and the annual revenue requirement for that last regulatory year, and
 - (3) Must conform with whichever of the following requirements is applicable:
 - (i) If the control mechanism relates generally to Standard Control Services the X factor must be designed to equalise (in terms of net present value) the revenue to be earned by us from the provision of Standard Control Services over the regulatory control period with our total revenue requirement for the regulatory control period
 - (ii) If there are separate control mechanisms for different Standard Control Services the X factor for each control mechanism must be designed to equalise (in terms of net present value) the revenue to be earned by us from the provision of Standard Control Services to which the control mechanism relates over the regulatory control period with the portion of our total revenue requirement for the regulatory control period attributable to those services.

(c) There may be different X factors:

- (1) For different regulatory years of the regulatory control period, and
- (2) If there are 2 or more control mechanisms for each control mechanism.

We note that clause 6.5.9(2) of the NER does not apply for the regulatory control period 2015-20 due to the operation of transitional arrangements currently in place for us. Nevertheless, we have aimed to align our X factors in such a way as to achieve this requirement to the extent practicable in the regulatory control period 2015-20.

The following section sets out how our X Factor values are calculated and are consistent with the requirements of clause 6.5.9 of the NER.

9.2 X Factor Calculations – Standard Control Services

In order to determine the X Factors for each year of the regulatory control period 2015-20, the 2014-15 annual revenue value first needed to be determined. The Maximum Allowable Revenue (MAR) value for 2014-15 (\$1750.8 million) was determined using the formula:

 $MAR_{2014-15} = AR_{2013-14} \times (1 + \Delta CPI_{2014-15}) \times (1 - X_{2014-15}) + S_{2014-15} + C_{2014-15} + Transitional_{2014-15} + Passthrough_{2014-15}$

Where:

- AR₂₀₁₃₋₁₄ = \$1,501.09 million as approved by the AER in 2013-14
- $(1+\Delta CPI_{2014-15}) = 1.0293$ based on the March to March CPI value for 8 cities published on the Australian Bureau of Statistics website on 23 April 2014
- $(1-X_{2014-15}) = 1.0187$ based on the X4 value in the merits review decision
- $S_{2014-15} = 31.48 million based on the result of the formula $S_{2014-15} = AR_{2014-15} \times (1 + S \text{ factor} in 2012-13) AR_{2014-15}$, an S factor of 2% in 2012-13 and $AR_{2014-15} = AR_{2013-14} \times (1 + \Delta CPI_{2014-15}) \times (1 X_{2014-15})$
- $C_{2014-15} =$ \$63.65 million transitional capital contribution under recovery as reported in the 2012-13 RIN
- Transitional₂₀₁₄₋₁₅ = \$2.34 million as reported in the 2012-13 RIN
- Passthrough₂₀₁₄₋₁₅ = \$84.02 million for the feed-in tariff cost pass through approved by the AER.

This is consistent with the MAR used in the 2014-15 annual pricing proposal approved by the AER.

The value of X1 to X4 were calculated in such a way as to allow us to achieve a targeted revenue outcome which comprises the ARR values as calculated by the PTRM and other revenue allowances (not determined using the PTRM) that we need to recover. The other revenue allowances include:

- STPIS
- DMIS Carryover from the current regulatory control period
- DUOS unders and overs
- transitional capital contribution unders and overs
- transitional shared assets unders and overs.

The table below sets out the smoothed ARR values as calculated by the PTRM and other revenue allowances (not determined using the PTRM) for 2014-15 and for each year of the regulatory control period 2015-20. The sum of these revenues gives the total 'targeted' revenue for each regulatory year, which represents the DUOS charges excluding the feed-in tariff allowance.

Table 34: Ergon Energy's annual 'targeted' revenue requirement for 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
ARR (smoothed)	1,137.71	1,522.33	1,709.11	1,712.72	1,716.33
<i>plus</i> Other revenue allowances (excluding the feed-in tariff cost Pass-through)	171.40	183.19	-	-	-
Targeted Revenue (excluding the feed-in tariff cost pass-through)	1,309.11	1,705.52	1,709.11	1,712.72	1,716.33

The table below sets out the components of the other revenue allowances (not determined using the PTRM) for each year of the regulatory control period 2015-20.

	2015-16	2016-17	2017-18	2018-19	2019-20
STPIS allowance	22.75	(7.56)	-	-	-
less DMIS carryover	-	(1.90)	-	-	-
plus DUOS unders and overs	67.10	-	-	-	-
<i>plus</i> Transitional capital contribution unders and overs	85.12	114.66	-	-	-
<i>less</i> Transitional shared assets unders and overs	(3.58)	(3.48)	-	-	-
<i>less</i> Cost pass throughs – Feed-in tariffs in 2013-14 and 2014-15	-	-	-	-	-
Total other revenue allowances	171.40	183.19	-	-	-

Table 35: Ergon Energy's other revenue allowances for 2015-20 (\$M nominal)

The nominal total targeted revenues were modelled such the following outcomes for customers could be achieved:

- The ARR (i.e. not including other revenue adjustments) for 2015-16 is equal to that in the AER's Preliminary Determination for 2015-16.
- The total targeted revenue in 2016-17 is set equal to 97.22% of the total targeted revenue in 2014-15 in nominal terms.
- The total targeted revenue in each year from 2015-16 to 2019-20 inclusive should not exceed the 2014-15 in nominal terms.
- The average change in the total targeted revenue from 2015-16 to 2019-20 inclusive should not exceed the forecast CPI of 2.55%.
- The annual percentage change in the total targeted revenue from 2016-17 to 2019-20 inclusive are set to be equal.
- The resulting ARR to be recovered from Standard Control Services (i.e. the total targeted revenue less the other revenue adjustments) over the regulatory control period 2015-20, must be NPV neutral compared to the unsmoothed building block revenue values calculated by the PTRM (consistent with clause 6.5.9(b)(3)(i) of the NER).
- The smoothed ARR in 2019-20 should not be less than the unsmoothed building block revenue calculated in the PTRM and should not exceed the unsmoothed building block revenue by more than 7.5% (this is consistent with clause 6.5.9(b)(2) of the NER which requires that X factors be set such as to minimise, as far as reasonably possible, variance between expected revenue for the last regulatory year of the regulatory control period, and the

annual revenue requirement for that last regulatory year). Ergon Energy's smoothed ARR in 2019-20 is only 6.15% higher than the unsmoothed ARR.

The total targeted revenue values in 2017-18 through to 2019-20 inclusive are calculated by multiplying the previous year's total targeted revenue value by CPI and by the annual percentage changes in the nominal total targeted revenues. The annual percentage changes in the nominal total targeted revenues. The annual percentage changes in the nominal total targeted revenues were modelled to achieve these outcomes by goal seeking the annual percentage change in the total targeted revenue from 2015-16 to 2016-17 such that the difference between the NPV of the Resulting ARR values and the NPV of the unsmoothed ARR values is zero. As noted above, the annual percentage changes from 2016-17 to 2019-20 inclusive are set to be equal.

The Resulting ARR values, in each regulatory year of the regulatory control period 2015-20, are calculated as the difference between the total targeted revenue for that year and the other revenue adjustments in that year.

The values of X1 to X4 are calculated based on the percentage change in the Resulting ARR between each regulatory year from 2016-17 to 2019-20 inclusive. Specifically:

X1 = (Resulting ARR for 2016-17) / (AER's Preliminary Determination ARR for 2015-16) / CPI

X2 = (Resulting ARR for 2017-18) / (Resulting ARR for 2016-17) / CPI

X3 = (Resulting ARR for 2018-19) / (Resulting ARR for 2017-18) / CPI

X4 = (Resulting ARR for 2019-20) / (Resulting ARR for 2018-19) / CPI

The table below sets out the resulting X factors (i.e. X1 to X4) calculated using this methodology (consistent with clause 6.5.9(c)(i) of the NER).

Table 36: Ergon Energy's X factor values calculated for 2015-20 (\$M nominal)

	2015-16	2016-17	2017-18	2018-19	2019-20
X Factors	36.63%	(30.48%)	(9.48%)	2.28%	2.28%

10 List of documents referenced

Reference	Title
03.01.02	(Revised) Other Revenue Adjustments
03.01.03	(Revised) Application of Incentives Schemes
03.02.02	(Revised) Proposed Application of STPIS
7.00.03	(Revised) Customer Connection Initiated Capital Works Expenditure Forecast Summary
04.01.00	(Revised) Compliance with Control Mechanisms
04.01.02	(Revised) Jurisdictional Schemes
04.01.05	(Revised) Control Mechanism Model
05.03.01	(Revised) Default Metering Services Summary
06.01.01	(Revised) Operating Forecast Expenditure Summary Document
N/A	Houston Kemp: Depreciation report